



# Living with inflammatory bowel disease: The experiences of adults of South Asian origin

**Short Title: The LISA Project** 

**FINAL REPORT** 

February 2015

**NACC Grant reference SP2012-1** 

Suzanne Mukherjee, Bryony Beresford, Shaji Sebastian and Karl Atkin

Copyright © University of York, 2015

All rights reserved. Reproduction of this report by photocopying or electronic means for non-commercial purposes is permitted. Otherwise, no part of this report may be reproduced, adapted, stored in a retrieval system or transmitted by any means, electronic, mechanical, photocopying, or otherwise without prior written permission of the Social Policy Research Unit, University of York.

ISBN 978-1-907265-32-7

A CIP catalogue record for this report is available from the British Library.

Further copies of this report or any other Social Policy Research Unit publication can be freely obtained by visiting our website:

#### www.york.ac.uk/spru

Paper copies can be obtained from:

Social Policy Research Unit University of York Heslington York YO10 5DD

# **Contents**

Acknowledgements	i
Contributions from the research team	iii
Executive Summary	v
Chapter 1 Introduction	1
Background to the project	1
Aims of the research	2
Overview of research design and methods	2
Structure of the report	3
Chapter 2 Background Literature	5
What do we know about the experience of living with IBD?	6
Current approaches to research on the experiences of ethnic minority group their support needs	
What do we know about the experiences of those who are South Asian?	9
Chapter 3 Research Design and Methods	13
Preliminary work	13
Recruitment and training of interviewers	13
Recruitment of research participants	14
Development of the interview topic guide	15
Data collection	16
Ethical considerations	17
Data analysis	18
The research participants	19
Chapter 4 Living with IBD: The Shared Experience	23
Introduction	23
Dealing with physical symptoms	23
Support from others	24
The emotional toll of IBD	25
Impact of IBD on everyday life	27
Social and leisure activities	27
Education, employment and finances	28
Relationships with partners and spouses	
Parenting and caring responsibilities	

Chapter 5 How Does South Asian Identity, Culture and Faith Influence the Experience of Living with IBD?	. 31
Introduction	. 31
Culturally specific difficulties understanding IBD	. 31
The stigma of ill health within South Asian communities	
The perception that bowel movements are a taboo subject	. 35
Beliefs about the links between food and health	. 35
Alternatives to conventional medical treatment	. 36
Cultural expectations regarding a woman's role within the family	. 37
Cultural expectations regarding a man's role	. 39
'Spicy' food is the norm	. 39
Living with extended family	. 40
Visits to extended family living overseas	. 45
Practising faith	. 46
Chapter 6 Experiences with Health Services	. 49
Introduction	. 49
Support from GP services	. 49
Support from gastroenterology services	. 50
Positive aspects of support provided by gastroenterology services	. 50
Difficulties with the support provided by gastroenterology services	. 53
Views on how well services attend to the needs of South Asian patients	. 57
Additional support wanted from health services	. 59
South Asian specific recommendations	. 60
Chapter 7 Views on Crohn's and Colitis UK and the Support it Provides	. 61
Introduction	
Awareness of Crohn's and Colitis UK	. 61
Reasons why individuals did not engage with Crohn's and Colitis UK	. 62
Do cultural issues impact on South Asian patients' engagement with Crohn's an Colitis UK?	
Barriers to use of on-line resources	. 64
The representation of South Asian individuals in publicity and information resources	. 64
Support groups	. 64
Support in languages other than English	. 65

Views on how Crohn's and Colitis UK can better support South Asian patients	66
Awareness raising about IBD in South Asian communities	66
Supporting the information needs of the wider family	66
Addressing specific information needs	67
Chapter 8 Discussion	69
Introduction	69
Summary of the findings and what they add to the literature	69
Implications for practice	74
Psychological Support	74
Information on the usefulness and safety of CAMs for IBD	75
Dietary advice which is culturally appropriate	75
Awareness raising about IBD within the South Asian communities and families	
The need for advice and support on making trips to visit family overseas	76
Increasing the visibility of South Asian members of Crohn's and Colitis UK	76
Strengths and limitations of the study	77
References 78	
Appendix 1 Interview Topic Guide	85
Appendix 2 Visual Aids for Interview	93
Appendix 3 The development of the thematic framework	95
Appendix 4Thematic framework	99

# **Acknowledgements**

This project was funded by a 'Living with IBD' research grant from Crohn's and Colitis UK (NACC Grant reference: SP 2012-1). We would like to thank the Board of Trustees for their interest and engagement with the project, and the organisation for the financial support provided. We are extremely grateful to the gastroenterology staff from the five research sites involved in the project who assisted with the recruitment of research participants. Their willingness to work alongside the research team to develop and implement an appropriate recruitment strategy was instrumental in this study achieving its recruitment target. We would also like to acknowledge members of the Project Advisory Group, who provided invaluable advice and encouragement to the research team throughout the research process. Administrative support to the project was provided by Teresa Frank and Sally Pulleyn, and we would like them both for their efficiency and attention to detail. This project would not have been possible without the contribution of our research participants, who gave up their time to talk to us about the experience of living with IBD. For many this involved sharing personal information and we would like to let these participants know how much we appreciate this. Finally, we would like to thank the LISA project interviewers for collaborating with us in the development of the interview topic guide and for working so hard to carry out sensitive in-depth interviews. These interviewers were: Zahira Latif, Melanie Lovatt, Hina Patel, Penny Rivlin, Lucy Stuttard, and Jim Wade.

### Contributions from the research team

### Dr Suzanne Mukherjee, SPRU, University of York – Principal Researcher.

Recruitment of project advisory group (PAG) members and running of PAG meetings; recruitment, training and supervision of LISA Project interviewers; coordination of research participant recruitment; development of recruitment and interview materials; pilot interviews; data analysis, including review of all interview transcripts, development of thematic framework, and charting, summarising and interpretation of data; write up of final report.

**Professor Bryony Beresford, SPRU, University of York – Co-applicant for research funding.** Attendance of PAG meetings; recruitment and training of LISA project interviewers; development of recruitment and interview materials; contribution to data analysis, including review of a selection of interview transcripts, development of thematic framework, and summarising a selection of charts; review of final report.

Professor Karl Atkin, The Department of Health Sciences, University of York - Co-applicant for research funding. Advice on ethnicity related issues throughout the project; attendance of PAG meetings; input to LISA Project interviewer training days; contribution to data analysis, including development of 'thematic framework' and review of chart summaries; review of final report.

**Dr Shaji Sebastian, Hull and East Yorkshire Hospitals NHS Trust - Co-applicant for research funding.** Advice on clinical issues throughout the project; recruitment of research sites; attendance of PAG meetings; input to LISA Project interviewer training day.





# **Executive Summary**

# Living with inflammatory bowel disease: The experiences of adults of South Asian origin

### **Background**

IBD is chronic inflammatory disorder with no known cure. There are two main forms: ulcerative colitis (UC) and Crohn's disease (CD). In both conditions, individuals experience periods of remission and recurrent attacks. Symptoms are highly variable depending on which part of the bowel is affected, but can include: diarrhoea, with blood and mucus; faecal incontinence; abdominal pain; weight loss; vomiting and fatigue. Current treatments aim to bring about remission, but can have unpleasant side effects. Even when well, fear of bowel related symptoms can result in people curtailing their daily activities, impacting on work, leisure and social life.

Globally IBD is most prevalent in Europe and North America, and in recent years there has been an increase in incidence in people of South Asian origin living in these countries. However, little is known about their experience of the condition. This is of concern because it has been known for some time that minority ethnic groups with a range of health conditions encounter barriers to accessing acceptable health and other support services.

# **The LISA Project**

The aim of the LISA Project was to describe what living with IBD is like for South Asian adults, and to understand whether ethnicity impacts on this experience and, if so, how. Thirty-three people with IBD (20 women, 13 men), recruited from five gastroenterology clinics in England, took part in an in depth face-to-face or telephone interview. The sample reflects the diversity of the South Asian population in the UK, including people who identified themselves as being Indian/British Indian; Pakistani/British Pakistani and Bangladeshi/British Bangladeshi. Participants also represented the three most prominent faith groups (Hindus, Muslims and Sikhs). Participants had been diagnosed with IBD for between three months and 21 years (median six years). Eighteen had been diagnosed with Crohn's disease, 14 with ulcerative colitis; and one person with indeterminate colitis.

# **Key findings**

#### Living with IBD - the shared experience

Findings suggest there is a generic experience of IBD, irrespective of ethnic background. In common with others who have IBD, participants reported that flare ups were painful, exhausting and involved frequent trips to the toilet. As a result many were housebound when experiencing them. In between flare ups, ongoing milder symptoms were reported.

The majority of participants adhered to prescribed medication, but also used a wide range of other strategies to control or cope with the symptoms of IBD. Practical and emotional support was typically provided by immediate family. Extended family were often not in a position to help due to living too far away or having caring, or other, responsibilities. Many interviewees described the emotional toll of having IBD. Feeling low, depressed or anxious added further to the difficulties people experienced so that, even when symptom free, some curtailed their activities due to loss of interest or worries about the possibility of experiencing symptoms. This combination of physical symptoms and emotional difficulties meant individuals struggled at times with maintaining a social life and friendships, completing higher education/training courses, managing work, caring for children, and relationships with partners. A few also reported financial difficulties.

#### South Asian influences on living with IBD

Although there was a 'generic' IBD experience, the research revealed that aspects of South Asian identity, cultures and/or faith did impact on the experience of living with IBD. These are described below.

#### Culturally specific difficulties understanding IBD

Participants said there were a number of culturally specific reasons why people within the South Asian community had difficulty understanding IBD. These include: many people have never heard of the condition; there is no word for Crohn's in some South Asian languages; and the word 'disease' has different connotations in South Asian communities.

#### A culture of silence around ill health and bowel symptoms

Most participants spoke of being reluctant to talk to others within the South Asian community about having IBD for two reasons. First, ill health is particularly stigmatised in South Asian communities due to the perception that it impacts negatively on 'marriageability' and caused unfounded concerns for the health of future offspring. Furthermore, while it was acknowledged that talking about IBD symptoms (particularly bowel-related) might be embarrassing for many people, there was a perception that this subject was particularly taboo within South Asian communities.

#### 'Spicy' food is the norm

Most participants believed that South Asian food exacerbated IBD symptoms and avoided it. This could mean eating, or preferring to eat, different food from other members of the household. This was manageable when the person was in charge of family meals and cooking, but more difficult if they were not (i.e. young person living with parents, woman living with in-laws). A reluctance to eat/avoidance of spicy food was also an issue when out socially, particularly when participants had chosen not to tell others they had IBD. Some dealt with this by avoiding social events. Occasionally the difficulty of refusing food, or social pressure to eat, either in or outside the home, meant that people ate despite knowing it would make them unwell.

#### Beliefs about the links between food and health

Some participants spoke of difficulty getting others to accept the diet they had chosen to follow because of cultural and religious beliefs about food. They spoke of two types of beliefs. First, the belief that health problems are the result of an 'imbalance' in the foods a person eats, resulting in the perception that might be possible to be cured of IBD through changing your diet. Second, beliefs within a particular faith that certain foods are blessed and therefore must be good for you.

#### Cultural expectations regarding roles within the family

Women, in particular, from any background might worry about their ability to manage childcare and housework when ill. The women participating in this study were furthered concerned about the social consequences of not being able to fulfil these roles, feeling that they would not be forgiven by others in their community if they failed to take on the duties expected of a wife and mother.

#### Visits to family living overseas

Making trips overseas to visit family was common and many worried about this due to concerns about travelling whilst experiencing bowel related symptoms and limited access to toilets. Many were pleased to find their health improved while away. Where people did experience flare ups, there were often clear reasons for this, such as not adhering to medication or changes to diet.

#### **Practising faith**

A number of participants spoke of deriving a great deal of benefit from their faith, describing ways in which it helped them to cope with having IBD. However, Muslim participants reported difficulties praying because it involved a physical act which they were sometimes not well enough to perform. Furthermore, when experiencing bowel related symptoms they often limited the amount of time spent away from home due to few public toilets having the amenities needed for them to clean themselves with water, as required by their faith.

#### **Experiences of gastroenterology services**

Experiences of gastroenterology services were overwhelmingly positive, with just a few exceptions. Participants appreciated the ease of access to the team outside of scheduled appointments; how comfortable they felt talking to staff about sensitive issues, such as bowel related symptoms; and the information provided about IBD. Nevertheless, there was disappointment that the focus of consultations was purely on medical treatment, with staff not having or taking the time to discuss how they were coping with the condition, failing to provide dietary advice, or respond to requests to see a dietician. Many were using, or interested in using, Complementary and Alternative Medicines (CAMs). However, a number had not discussed these options with the gastroenterology clinic team feeling that they were not interested in, or 'negative' about, such treatments. Very few people consulted their GP for advice or treatment for IBD.

When asked about their needs as South Asian patients, some interviewees offered experiences of culturally sensitive practice (i.e. consultants taking time to help prepare people for trips overseas, advising patients about the increased risk of vitamin D deficiency). Where shortcomings did exist, a lack of adequate translation services was mentioned, as well as staff not appreciating the importance to (some) South Asian patients of being able to fulfil cultural expectations around what constitutes a 'good' wife or mother, and the need to find treatments and/or strategies which enabled them to take on these roles.

#### **Support from Crohn's and Colitis UK**

Knowledge and understanding of Crohn's and Colitis UK was typically very low. Key reasons why people did not engage with the organisation were:

- the episodic nature of the condition, with people reporting not being motivated to use the organisation when well and being too ill to do so when having a flare up;
- many felt they already had all the information and support they needed, largely through gastroenterology clinics, family and friends;

• the organisation does not provide the 'uncensored' information about dietary regimes and CAMs which, although untested, were of interest.

A number of South Asian specific barriers to engagement with Crohn's and Colitis UK were also identified. It was suggested that the older generation of South Asian patients are unlikely to be 'on line' and/or necessarily able to read English, making it hard for them to access Crohn's and Colitis UK's website. Barriers to attending meetings of local Crohn's and Colitis UK groups included:

- the belief that they would be the only South Asian person present and this would feel uncomfortable:
- the use of culturally inappropriate venues such (i.e. pubs);
- a perception that South Asian people are relatively private so would be uncomfortable sharing experiences in a group.

Finally, the fact that most of Crohn's and Colitis UK publications are only available in English, and telephone helplines are unlikely to be manned by people who speak South Asian languages, were highlighted as shortcomings that might need to be addressed.

#### Recommendations

South Asian patients are a diverse group, differing considerably in their faith and cultural backgrounds, as well as in their social and economic circumstances. An individualised approach to care is therefore needed. That said, the findings from the LISA Project suggest that there are a number of things that could be done to improve the life experience of South Asian people living in the UK who have IBD. Some of the recommendations made are generic. That is, developing support would benefit both the South Asian and the wider IBD population. Other recommendations only apply to those who are South Asian. Addressing these issues may require new, innovative partnerships between organisations and agencies concerned with the health and well-being of South Asian adults in the UK living with IBD.

# Action that would benefit those who are South Asian <u>and</u> the wider IBD population

- Increasing access to psychological support. This should include:
  - opportunities for people to discuss how they are managing their everyday life, and any strategies that might be helpful;
  - access to psychological therapies for those experiencing feelings of depression and/or anxiety brought on or exacerbated by having IBD.
- Providing advice on the usefulness and safety of Complementary and Alternative Medicines (CAMs). Our findings suggest that without this patients will use unofficial online forums for information and advice on treatments and remedies, and may conceal this from their consultant/other clinic staff.
- **Greater information and advice on diet**, including access to a dietician for individualised support.
- Offering advice and support on making trips overseas, including:
  - o the storage of medication when travelling;
  - how best to deal with a flare up should this occur while overseas;
  - reminders as to the importance of adhering to prescribed medication and usual diet;
  - a brief written summary of the person's diagnosis and treatment in case there is a need for medical attention.

#### Action aimed specifically at supporting people with IBD who are South Asian

- Increasing awareness and understanding of IBD within South Asian Communities. Participants suggested that Crohn's and Colitis UK could assist with this by: developing culturally appropriate information resources; ensuring these are available in South Asian languages; and visiting temples and mosques to distribute such material.
  - Gastroenterology clinics can also play a role, increasing understanding and awareness, particularly among close family members. For example, asking patients if they would welcome the opportunity for family members to 'sit in' on an appointment or meet with the gastroenterology team.
- **Dietary advice needs to be culturally appropriate**, including advice on foods typically eaten by South Asian families, as well as on the risks and benefits of fasting.
- Increasing the visibility of people of South Asian origin on Crohn's and Colitis UK materials, and their representation within the organisation at both a local and national level.

# **Chapter 1** Introduction

### **Background to the project**

Although Inflammatory Bowel Disease (IBD) has traditionally been considered a condition prevalent in white European populations and those of Jewish origin, in recent years an increased incidence has been reported in people of South Asian (South Asian) origin who live in Western countries (Molodecky et al., 2012). The rise in incidence of IBD in this group is of particular concern in the UK, where efforts are being made to understand why this pattern may be emerging (Ravikumara and Sandhu, 2006; Wilson, 2011). However, little attention has been given to the experiences of South Asian patients living with IBD, despite guidelines for health professionals on the management of IBD acknowledging that the patient's perspective and expectations should be taken into account (Mowat et al., 2011); and concern that the South Asian population is not making use of self-help groups provided by Crohn's and Colitis UK (Moody et al., 1993).

In the UK it has been recognised for some time that minority ethnic groups with a range of health conditions experience barriers to accessing acceptable health and other services. Recent research into conditions such as diabetes, coronary heart disease and various forms of cancer suggest that difficulty accessing appropriate services continues to be an issue for South Asian patients. In addition to social disadvantage and social exclusion, difficulty is often explained by misperceptions among service providers about the needs of this group (Kalrsen et al., 2012, Postenote, 2007; Szczepura, 2005). Such findings emphasise that the importance of understanding when ethnicity and cultural differences make a difference to a person's experience and when they do not.

Most research on the illness experience of IBD among South Asian people in the UK is based on studies carried out some years ago, predominantly with people of Indian origin, which focused on specific issues such as diet, use of complementary medicine, and information needs (Moody et al., 1998; Probert et al., 1996; Conroy and Mayberry, 2001). In recent years a study has been carried out which took a more comprehensive approach, investigating the lived experiences of people from South Asian communities (Nash et al., 2011). However, this research focused specifically on young people aged 16-24 years, and was based on a small sample (n=20), most of whom were Muslim. It is essential that research on the experiences of the South Asian patients is up to date since the experiences of this population are changing rapidly (Jayaweera, 2011), and reflects and explores the diversity of the South Asian population, in terms of country of origin (and/or association), faith and religion, socio-economic circumstances, history of migration and whether individuals are UK born (Craig et al., 2012). Therefore, there is still a significant gap in the

research literature, with the experiences and views of young people from other faith backgrounds, as well as older South Asian IBD patients, left unexplored. It was against this background that Crohn's and Colitis UK funded the LISA Project - a research project investigating the experiences of South Asian adults in the UK living with IBD.

#### Aims of the research

The primary aim of this study was to describe what living with IBD is like for South Asian adults and to understand whether ethnicity impacts on this experience and, if so, how. In addition, this study aimed to identify met and unmet needs for support. The specific research objectives were to:

- understand the impact of IBD on everyday life for people from a diverse range of South Asian communities;
- explore perceptions and understanding of the condition and how this influences lifestyle/treatment decisions;
- describe the support (both formal and informal) that members of the South Asian community have available to them and their feelings about utilising this support;
- to understand and explain the extent to which ethnicity and broader cultural and religious difference moderates the experience of IBD, while also exploring the extent to which there is a generic experience of IBD, irrespective of ethnicity;
- to offer Crohn's and Colitis UK evidence to inform and support the way it seeks to better respond to the needs of the South Asian population living with IBD.

# Overview of research design and methods

As this was the first research to be carried out on this topic, the study adopted an exploratory approach aimed at understanding the potentially diverse experiences of the South Asian population, without presupposing what those experiences might be. In-depth qualitative interviews were used to gain an insight into participants' perceptions, experiences and support needs. Qualitative interviews are particularly effective in exploring complex and sensitive issues, and in understanding people's belief systems and how they interpret what is happening to them.

Given that many people with IBD are reluctant to talk about their condition to anyone other than close friends and family (Kelly, 1992), and that the response rate of IBD patients from the South Asian community to other studies has been found to be substantially lower than that of European patients (Moody et al., 1993, 1998; Conroy and Mayberry, 2001), careful thought was given to how to overcome these potential barriers to recruitment. A key element was providing participants with a choice of interviewer. Specifically, participants were aware, when recruited to the study, that they could state a preference regarding the gender of their interviewer, whether or not he/she has IBD, their ethnicity (that is, South Asian or white British) and which

language they were interviewed in. The research team would endeavour to match them with an interviewer who met their requirements. In addition, participants were able to choose between a face-to-face interview (at home or in hospital) or a telephone interview. Such an approach has been used in other fields of research where the subject matter for discussion is sensitive and/or being interviewed by someone with a 'shared experience' is likely to support both recruitment to the study and openness in the interview (Warr et al., 2011; Beresford et al., 2008). In order to be able to offer this choice of interviewer, qualitative researchers already working in the department undertaking the research were supplemented by a small number of freelance qualitative researchers specifically recruited and trained for this project.

Research participants were recruited through gastroenterology clinics located in different areas of England. A purposive sampling strategy was adopted to ensure that men and women, ulcerative colitis and Crohn's disease patients, and a range of ages were represented. Special efforts were made to ensure the sample reflected the complex heterogeneity and inter-relation of regional, religious, linguistic, and national identities within the South Asian population (Karlsen and Nazroo, 2006).

## Structure of the report

In Chapter 2 of this report, we put the research into context, outlining what is already known about the experience of living with IBD, the specific experiences of South Asian patients, why it is important to look at the experiences of ethnic minority groups and how best to approach such research. Chapter 3 provides details on the research design and methods, including the characteristics of the final sample of patients who took part in the study. The research findings are reported in Chapters 4-8. Chapter 4 reports on the generic experience of IBD – that is experiences that South Asian patients share with the wider IBD patient population. In Chapter 5 we go on to describe findings on how South Asian culture and faith influence the experience of living with IBD. Chapter 6 describes patients' experiences with health services, including both GP and gastroenterology services, while Chapter 7 deals with participants' views on Crohn's and Colitis UK and the support it provides. Finally, in Chapter 9 we reflect on the research, discussing what the study adds to the existing literature on the experiences and support needs of adults with IBD, and the implications of the findings for Crohn's and Colitis UK and health service providers.

# **Chapter 2** Background Literature

#### Introduction

In this chapter, for readers who are unfamiliar with IBD, we begin by providing an overview of the condition and its treatment. Then, in order to help contextualise the findings from the LISA project, the generic evidence on what it is like to live with IBD, and the experiences of the South Asian patient population, are reported.

## **Understanding IBD and its treatment**

IBD is chronic inflammatory disorder with no known cure. There are two main forms: ulcerative colitis (UC) and Crohn's disease (CD). A recent review of epidemiological data across the world indicate that that IBD is most prevalent in Europe and North America (Molodecky et al., 2012). Based on estimates from this review it is suggested that up to 620, 000 people in the UK may currently be living with IBD (IBD standards update, 2013).

In both conditions the clinical course is variable with individuals experiencing periods of remission and recurrent attacks. This may result in frequent hospital visits and spells as an inpatient. Symptoms include diarrhoea, with blood and mucus, abdominal pain, weight loss, and sometimes vomiting. Fatigue is often prominent and can be disabling. Faecal incontinence is also common. Possible complications include anaemia, malnutrition, intestinal obstructions, abscess formation and fistulas. The exact nature of symptoms experienced depends on which part of the bowel is affected, resulting in considerable variability between patients (Ford et al., 2013; Kalla et al., 2014; Norton et al., 2013; Czuber-Dochan et al., 2013; Thompson, 1990). In woman with CD fertility is reduced, but evidence suggests this is linked to voluntary childlessness, rather than there being any physiological causes. In men with CD there is also a reduction in fertility, but not in reproductive capacity (Tavernier et al., 2013). While some people may appear ill because of the bloating effects of medication or, alternatively, because they are thin, pale, and emaciated, others may not appear in any way different and can choose to keep their health difficulties private if they wish. It has been suggested that many people opt to conceal the nature of their illness to all but a few intimate friends and or relatives because of the embarrassing nature of some of their symptoms (Kelly, 1992).

Current medical treatment aims to bring about a remission in the condition and, when this has been achieved, prevent relapse thereby minimising the impact of IBD on quality of life. Treatments for both UC and CD include: drug therapy, to reduce symptoms, inflammation and suppress the immune response; nutrition, including an individualised diet and, in some, enteral nutrition; and surgery. For patients with UC surgery can be curative but may mean having an ileostomy or ileoanal 'pouch'.

Surgery may also be offered to patients with CD, but here it largely alleviates symptoms for a period of time, and the condition is likely to recur in other sections of the digestive tract (Ford et al., 2013; Kalla et al., 2014). While drug therapies are increasingly effective, resulting in a reduction in the use of surgery (Ford et al., 2013), the side effects of such therapies can be unpleasant and of concern to patients (Hall et al, 2007). Long-term use of corticosteroids can also lead to the development of osteoporosis and therefore requires careful management. Both UC and CD patients are now considered at increased risk of colorectal cancer and, in the case of Crohn's, there is also a small increased risk of bowel cancer. In the UK, colonscopic surveillance is recommended for patients starting 10 years after the onset of symptoms, with the exception of those with ulcerative proctitis (Cairns et al., 2010). Investigations carried out in order to make an accurate diagnosis, decisions about management of the condition, and for cancer surveillance can be uncomfortable, involving various forms of endoscopy, barium x-rays and scans. In summary, while progress in the diagnosis and treatment of IBD has resulted in marked improvements in prognosis for people who have the condition, it remains the case that IBD can have a considerable impact on those who have the condition.

## What do we know about the experience of living with IBD?

Evidence as to the impact IBD has on the everyday lives of people living with the condition is provided by research using a number of different methodologies, and studies have explored a range of issues, including education and employment, mental health difficulties, quality of life, and the lived experience.

Research on the education and employment status of people with IBD does tend to suggest that there is an economic burden for patients as, despite attaining a similar level of education as the general population, there a is higher rate of non-participation in the labour force (Marri and Buchmann, 2005). When employed, people with IBD are said to have a high work ethic, with a recent UK survey finding that 80 per cent continue working despite feeling unwell (Gay et al., 2011). Nevertheless, or perhaps because of the effort involved in working while unwell, people often take early retirement and cite IBD as playing a major part in this decision (Crohn's and Colitis UK, 2014).

For many years there has been interest in whether people with IBD are at increased risk of mental health difficulties, initially driven by the belief amongst both patients and clinicians that IBD is exacerbated by stress (Theis and Boyko, 1994; Drossman, 1988). In recent years, a number of studies have been published indicating that rates of anxiety and depression are higher in people with IBD than in those without the condition (Goodhand et al., 2012; Loftus et al., 2011; Graff et al., 2009; Fuller-Thomson and Sulma, 2006; Kurina et al., 2001). It is, however, hard to come to any firm conclusions about prevalence rates since there is considerable variability

between studies in the rates reported. Importantly, it has now been established that factors such as active/severe disease and socio-economic status increase the risk of experiencing such difficulties (Hauser et al., 2011; Graff et al., 2009; Nahon et al., 2012). In other words, mental health difficulties are not inevitable but depend on individual circumstances.

Looking at the lives of people with IBD more generally, early research using generic quality of life measures suggested that overall people with IBD were doing well relative to other groups of patients (those with rheumatoid arthritis, chronic back pain, or receiving haemodialysis). However, this research also found that the impact of IBD is quite distinctive in that, unlike other chronic conditions, it is the social and emotional aspects of everyday life that are affected rather than physical functioning (Drossman et al., 1989; Patrick and Deyo, 1989; Hjortswang et al., 1999; Drossman, et al, 1991). As a result, researchers shifted to examining quality of life using IBD specific measures, one of the most popular being the Inflammatory Bowel Disease Questionnaire which assesses bowel, systemic, social and emotional functioning. Research using this measure has revealed variability in the impact of IBD on quality of life, with IBD symptomatology, higher perceived stress, lower socio-economic status, and low perceived social support all associated with lower scores. However, the direction of the relationship between these variables is often unclear (Sainsbury and Heatley, 2005; Moradkhani et al., 2013; Borgaonkar and Irvine, 2000).

Evidence that physical symptoms alone cannot explain patients' subjective sense of wellbeing, as well as a desire to understand patients' support needs, has led many to investigate the worries and concerns of IBD patients. An initial review of the research on the topic suggested eight major issues for patients: lack of energy; the sense of lack of control; body image (which can be altered by medication and surgery); isolation and fear (due to the embarrassing and unpredictable nature of the condition, and the possibility of developing colon or bowel cancer); feeling a burden to others; feeling dirty; and lack of information from the medical community, in part due to the cause of IBD being unknown (Casati et al., 2000). After taking perceived health status into account, women are reported to be more concerned than men about attractiveness, feeling alone and having children (Maunder and Esplen, 1999). Research comparing studies carried out in different countries has found crosscultural variation, both in the issues that concern patients and the level of concern reported (Hjortswang et al., 1999; Levenstein et al., 2002). However, the reasons for these differences are unclear, with some speculating that they may reflect differences in the socio-economic background or disease type of participants, rather than cultural differences per se (Sainsbury and Heatley, 2005).

There are concerns that the growing body of research on the psychological wellbeing and quality of life of patients with IBD does not fully represent the subjective experience of living with IBD and is too negatively focussed (Wolfe and Sirois, 2008). A recent review of research articles published up until August 2010 identified just six studies on the lived experience (Kemp et al., 2012). After carrying out a metasynthesis of the findings, the authors conclude that fear of incontinence, and behaviours related to this fear is the most significant issue to emerge from the data, resulting in people curtailing their daily activities and impacting on work, social and leisure functioning, even when well. They also note that trying to control the illness or symptoms was a theme in all studies and suggest that attempting to cope with the condition is all encompassing, so that that people often 'live in the world of the disease' even when in remission.

Qualitative research looking specifically at how people manage IBD indicates that many people show a determination to live as normal a life as possible and use a combination of different strategies to control or fight their disease (Hall et al., 2005). These include psychological (e.g. positive attitude, denial, acceptance etc.); behavioural (e.g. diet, situational avoidance, meditation, exercise, information seeking etc.); social (e.g. seeking support from others, being secretive, keeping up appearances); and biomedical (e.g. seeking medication, health care or surgery) strategies. Some strategies seem to be specific to IBD, or at least to bowel related conditions. For example, active management of one's diet, 'bathroom mapping' (where people make themselves familiar with toilet facilities available at places they frequent regularly or on a journey they make routinely); and avoiding locations and social functions without toilet facilities (McCormick et al., 2012). It has also been noted that some of the strategies people adopt overlap with the impact of the condition on their everyday life. For example, planning outings around toilet facilities can result in social avoidance and isolation (Hall, 2005).

# Current approaches to research on the experiences of ethnic minority groups and their support needs

As mentioned in Chapter 1, there is now a considerable body of literature highlighting the need for research on the illness experience of people from ethnic minority groups due to evidence that these groups experience health disadvantage as assessed by indicators of morbidity and mortality (Kalrsen et al., 2012; Postenote, 2007; Szczepura, 2005). In recent years the UK Department of Health has commissioned a number of initiatives in an attempt to ensure greater equality in health and social care provision (Postnote, 2007). Such initiatives are based on an understanding that provision of services alone does not ensure equitable access for all and that, in addition to making services accessible to patients by addressing their language needs, organisations and their staff require 'cultural competence' (Szczepura, 2005). However, unlike the United States where a number of states have introduced mandatory cultural competence training for health professionals, there is no such legislation in the UK. It therefore seems likely that some providers and practitioners will unfamiliar with the concept.

Cultural competence entails: understanding the importance of social and cultural influences in patients' health beliefs and behaviours, including their attitudes to care providers, and ensuring these influences do not become the basis of discrimination and inequality; considering any organisational, structural or clinical barriers to accessing care; and devising interventions that take such issues into account (Szczepura, 2005; Betancourt et al., 2003). Examples of organisational barriers include the extent to which the leadership of an organisation and its workforce reflect the ethnic composition of the patient group. Structural barriers include shortcomings such as lack of interpreter services or culturally/linguistically appropriate information, and slow speed of referral to services. Clinical barriers refer to socio-cultural differences between the patient and provider in relation to health beliefs, attitudes towards medical or other care, and levels of trust in health and other professionals. Even when service providers and patients are from the same ethnic background there may be differences in beliefs and attitudes.

There is, however, a danger that providing general information on the attitudes, values, beliefs, and behaviours of specific groups, in an attempt to increase the provision of culturally competent care, can in itself lead to further stereotyping. Therefore it is currently advised that, when discussing ethnic differences, it is important to stress that these are only potential differences. Service providers caring for people from an ethnic background should be encouraged to explore individual patients' perceptions of their condition, as well as the role that family, gender, and socio-economic issues play in their lives and service provision preferences (Betancroft, 2003). Recent research on the illness experiences of ethnic minority groups suggests that while some differences between groups are due to cultural influences, it is important not to overemphasise these as age, gender, social class and disadvantage can be equally important in understanding an individual's experiences. Individual's experiences also occur within a social context so that the meaning of the illness is negotiated through interactions with family members, people within their wider community, and health professionals. In effect, an individual's perception of who they are influences how they manage their illness, but others perceptions are equally important and will also influence their experience (Karlsen et al., 2012). It follows that anyone carrying out 'ethnicity research' needs to attend to this multitude of factors, and the potentially complex relationship between them, rather than simply viewing all experiences in relation to ethnic identity.

# What do we know about the experiences of those who are South Asian?

In general, research evidence on the experiences of people with IBD who are from an ethnic minority group is extremely limited. Ethnic minority groups are often poorly represented in studies of the wider IBD population. For example, in their recent metasynthesis of studies on the lived experience of people with IBD, Kemp et al., (2012) pointed out that, of the six studies included, only one had included anybody from an ethnic minority background (Kemp et al., 2012). In some cases researchers report excluding people whose first language is not English from studies due to the lack of resources to facilitate involvement (for example, Rochelle and Fidler, 2013; Borgaonkar et al., 2002).

Very few researchers have carried out research specifically investigating the experiences of people from an ethnic minority background. However, in the UK there is a small body of evidence on the experiences of South Asian patients in the Leicester area, based on studies carried out by the same group of researchers over a number of years. They began with a study investigating the characteristics of people who joined a newly formed Crohn's and Colitis UK support group, finding that those from ethnic minority groups were unrepresented. They warned that, not only might attending be beneficial for patients, but that their underrepresentation means that the needs of ethnic minority patients are frequently excluded from the plans made by the organisation (Moody et al., 1993). Next, in a postal survey examining the diet of South Asian patients with IBD they found that Hindu UC patients had significantly altered their traditional diet, reducing their intake of spices, flour and milk (Probert, 1996). In 1998 the group went on to research the use of complementary medicine in both European and Asian patients (Moody et al., 1998). They acknowledged that some UK hospitals offer complementary services (e.g. acupuncture, hypnotherapy, reflexology, osteopathy and herbal medicine etc.), but speculated that in communities with a substantial ethnic population other options such as Ayurvedic and Unani medicine, as well a traditional healers, may be more popular. A survey asking about advice sought from alternative practitioners was mailed out to 192 European and 190 Asian patients with IBD. Unfortunately, while a response rate of 83 per cent was achieved for European patients, only 42 per cent of South Asian patient returned the questionnaire. Results indicated that there was no significant difference in the use of complementary medicine between the two groups. However, there did seem to be a slight difference in the type of practitioner consulted. While the most commonly reported practitioners (herbalists and homeopaths) were consulted by both groups, only Asian patients had sought the help of a Hakim. Both groups reported consulting a spiritualist, but there did seem to be a greater tendency for South Asian patients to do so. South Asian patients were also significantly more likely to have found the alternative treatment beneficial. The authors note that they were surprised that South Asian respondents did not report the use of Ayurvedic practitioners, acknowledging that this may indicate that study suffered from sampling bias.

In 2001 the group carried out a survey of South Asian UC patients to obtain feedback on information leaflets on UC (Conroy and Mayberry, 2001). Forty people

<sup>-</sup>

<sup>&</sup>lt;sup>1</sup> A Hakim is physician using traditional remedies from a predominantly Muslim culture.

responded (representing a 53% response rate). Although 14 requested a leaflet in English, a further 21 requested it in Gujarati and 5 in Punjabi, indicating a clear need for information in South Asian languages. Almost all respondents felt that doctors should do more to help with language problems. This study was carried out some years ago and it is possible that the support health service offer to IBD patients whose first language is not English has changed in this time. The authors caution that although a third of respondents said the leaflet had alleviated anxiety, a third also reported it increased anxiety, so care must be taken to distribute such leaflets to those most likely to benefit from them and to be prepared for the fact that people may still need one to one discussions with health staff.

Aside from research undertaken by this group, there is one further qualitative study on the experiences of South Asian people with IBD in the UK. This investigated the experiences of 20 young people aged 16-24 years from an ethnic minority group, 18 of whom were South Asian and the majority Muslim (Nash et al., 2011). This found that young people of South Asian background faced three main challenges. First, where parents had limited English, the lack of translated information impacted on their ability to understand IBD and reduced parents' abilities to support their child either directly, or to act as an advocate for them both in health and education settings. Second, two-thirds of the sample had problems tolerating spicy food. This resulted in people sometimes feeling excluded from family life or, where separate food was cooked for them, guilty that they had created extra work for family members. It also made it difficult for them to participate in community life since this involved sharing food. In some cases the young people had chosen not to travel abroad to visit family because of lack of choice over the food they would be expected to eat. Third, young Muslim men reported some difficulty attending the mosque for prayers when experiencing symptoms due to the need for cleanliness. However, they derived strength from their faith as it helped them to stay calm and positive, and also provided an explanation as to why they were ill. Finally, participants also spoke of their experiences with health services, with two-thirds of the sample reporting significant delays in having their IBD diagnosed as a result of lack of referral by their GP. None attributed these delays to their ethnic background. However the authors note that in a few cases people were misdiagnosed with tuberculosis and a tropical disease, which they suggest indicates 'culturally diagnostic stereotyping'.

# **Chapter 3** Research Design and Methods

## **Preliminary work**

There were two main elements to preliminary work undertaken at the start of the project. First, a Project Advisory Group (PAG) was established. The group had 13 members, including patient representatives, gastroenterologists, IBD Specialist Nurses, representatives from Crohn's and Colitis UK, and an academic specialising in ethnicity research. The group met at two points during the project: prior to and on completion of field work. Issues discussed included: the strategy for recruiting research participants; the interview topic guide; preliminary research findings; and the potential implication of the findings for Crohn's and Colitis UK.

Second, visits were made to five gastroenterology clinics who had expressed an interest in being a research site for the study. During these visits the lead researcher meet with staff to: explain the study further; hear about staffs' experiences in caring for South Asian patients and issues they would like explored during the research; find out about the potential pool of research participants available from the site; and discuss the best approach to recruiting patients.

# Recruitment and training of interviewers

As outlined in Chapter One, in order to offer participants a choice of interviewer with respect to a number of key characteristics, it was decided that a team of 'in-house' (University of York) and freelance researchers would be used to carry out interviews. The posts for freelance interviewers were advertised through the national press and University websites, and applications went through a formal interview process. A team of six LISA project interviewers were recruited to the project (four freelance and two 'in-house'). All had an applied health or social science background and experience of conducting qualitative research on sensitive topics. Five were female and one was male. Two people had personal experience of living with IBD and two were of South Asian background. One of the South Asian interviewers was fluent in a range of South Asian languages (Hindi, Mipuri, Punjabi and Urdu); the other could speak a limited amount of Gujarati.

All the interviewers (freelance and 'in-house') attended a two day residential training course run by the research team. Input was also provided by a gastroenterologist and a member of the PAG who had a South Asian background and personal experience of living with IBD. During the course, interviewers were given an overview of IBD (including etiology, symptoms, investigations, treatment options) and

doing research with ethnic minority groups. In addition, specific training and preparation for the project was delivered including: arrangements for setting up and carrying out interviews; informed consent and data protection; the research topic guide; specific interviewing techniques, including differences between telephone versus face-to-face interviews; establishing personal boundaries; ethical issues likely to arise during fieldwork; and arrangements for the monitoring and supervision of work.

## Recruitment of research participants

Patients who took part in the study were recruited through five gastroenterology clinics in England. These clinics were purposively selected so as to include clinics with small and large proportions of South Asian patients and therefore different levels of experiences in supporting the needs of this patient group. Within these clinics the research team worked with nurses to identify all patients eligible for the study.

In order to be eligible for the study, patients were required to have:

- A confirmed diagnosis of IBD<sup>2</sup>;
- Be 18 years or older;
- Identify themselves as being of South Asian origin (come from or descend from a South Asian country, i.e. Bangladesh, India, Pakistan or Sri Lanka).

To protect patient confidentiality, the project was introduced to patients by clinic staff, who also passed on a project information pack (containing a project information leaflet, letter of invitation, response form and reply paid envelope). This project information pack was passed on either face to face (in clinic), or by post, depending on what staff at the gastroenterology felt was appropriate for their clinic and the circumstances of individual patients. Where possible, clinic staff went through the information pack with a patient, explaining its contents. Clinic staff were asked to make it clear to patients that they were under no obligation to participate in the project and that their decision would not affect the services they receive in any way. Project information packs were translated into appropriate languages (Bengali, Gujarati, Hindi, Punjabi, and Urdu) and made available to patients as required. In addition, arrangements were made for gastroenterology staff to make use of interpreters to explain the contents of the project information pack as and when necessary. If a patient was interested in taking part in the project, they were asked to return a response form to the research team, providing details on the type of person they would prefer to be interviewed by. Options included someone: male;

\_

<sup>&</sup>lt;sup>2</sup> Please note, our initial intention was only to include people who had been diagnosed for at least a year. However, during the early stages of fieldwork, interviews were carried out with people who had only recently been diagnosed. As they had a great deal to say about living with IBD, it was decided their views should be included in the study.

female; South Asian; around my age; with personal experience of IBD. Alternatively participants could indicate that they had 'no preference'. Participants were also asked which languages they were happy to be interviewed in. Language options were: English; Urdu; Gujarati; Hindi; Mirpuri; Bengali; and Punjabi. On receipt of a response form, the lead researcher allocated the interview to the interviewer who best matched the participant's preferences and this interviewer made contact with the potential participant to discuss the project and, if agreeable, make arrangements for an interview. Consent to take part in the project was obtained at the time of the interview.

A total of 101 people received an information pack about the project through their gastroenterology clinic, with 41 responding expressing an interest in the research (response rate 40.6 per cent). A further three members of PAG were invited to take part and two people approached the research team directly, having heard about the project through family or friends. In order to reflect the diversity of the South Asian population, out of the potential pool of 46 participants, a purposive sample of 37 people were selected to take part in the research and arrangements were made for the interview. Where a potential participant was not selected for interview, they were sent a letter thanking them for their interest in the project and letting them know that a short report on the project findings would be sent to them in completion of the project.

Out of the 37 selected for interview, three later decided not to take part in the project. In one case this was because of family concerns that the interview would be demanding for the person, who they perceived to be too elderly and unwell to take part in the process. In the remaining two cases, the decision was made by the patient themselves. In one case the reason given for delaying was that the person was experiencing a flare up; in the other the person was well and felt they would have little to contribute. One person had to be excluded from the study because further discussion revealed that his heritage was not South Asian as first thought. This resulted in a final sample of 33 patients. A description of the sample is provided at the end of this chapter.

# Development of the interview topic guide

The development of the interview topic guide involved a number of stages. First, an initial topic guide was drafted after: (a) reviewing the existing research literature on the lived experiences of patients with IBD and previous studies describing the experiences of South Asian patients with other health conditions; (b) consulting with gastroenterology staff as to their experiences in caring for South Asian patients; (c) presentation of potential interview topics to the PAG for review and comment.

Next the draft topic guide was presented to the LISA Project interviewers at their two day training course. Together the research team and interviewers reviewed the topic guide, discussing how best to approach potentially sensitive topics and whether the draft topic guide was missing any important issues. During this process the research team paid particular attention to the views of interviewers who had a South Asian background (n=2) or who had personal experience of IBD (n=2).

Following this discussion, the interview topic guide was amended and tested out by the lead researcher through pilot interviews (n=3). Feedback from these pilot interviews was positive, so only very minor changes were required, before sending a final version of the topic guide to the interview team for use.

#### **Data collection**

Each participant was interviewed once, with interviews lasting between 30 minutes and three hours. After discussions with the interviewer, 15 participants opted to be interviewed by telephone, 11 at home and seven in hospital (in a private room within or near to the gastroenterology department the patient attended). In terms of interviewer characteristics, 15 people were interviewed by a female interviewer with a South Asian background and 10 people by a female interviewer with personal experience of IBD. The remainder of the participants did not express a preference for a particular type of interviewer and were interviewed by a female white British interviewer. To thank people for their contribution to the project, all interviewees were offered a £20 shopping voucher, or could make a £20 donation to a charity of their choice.

Interviews began with the interviewer introducing themselves, explaining the project and taking informed consent. The interview explored: biographical details and initial diagnosis experience as a way of building rapport and contextualising subsequent discussions; diagnosis, management and the reactions of others to IBD; the impact of IBD on everyday life; experiences with health services; support from Crohn's and Colitis UK; ideas and recommendations for improving support to people of South Asian origin and their families. A copy of the interview topic guide is provided in Appendix 1.

In order to facilitate discussions two visual aids were used: (1) a 'life map' for use when discussing 'Living with IBD', particularly in cases where people found it difficult to spontaneously identify/talk about the impact of IBD on their lives; (2) a 'map' of the services/support provided by Crohn's and Colitis UK for use when discussing support from Crohn's and Colitis UK (please see Appendix 2 for the visual aids). The consent form and visual aids used during the interview process were available to participants in five South Asian languages.

In order to allow issues concerned with ethnicity to emerge from the interview, rather than presuppose what they might be, a staged approach was taken to the interviews, with initial questions encouraging participants to describe experiences in a general way. Only after this data was gathered were participants asked to reflect on the extent to which the factors such as faith, culture, age and gender impacted on their experiences.

In order to ensure data quality, the lead researcher reviewed all interviews shortly after completion. Reviewing interview material at this point allowed for an iterative data collection process, enabling the lead researcher to identify issues which required further exploration in later interviews, and who should be interviewed next (i.e. more males/females; more people from a particular faith) in order to reach 'data saturation'. Any issues that required further exploration were communicated to the whole of the interview team via email.

#### **Ethical considerations**

The fact that the focus of this research was on living with IBD, and that many people with this condition prefer not to talk about it with anyone other than close friends and family, meant that the topics discussed during interview were sensitive. In addition, it was possible, that in providing a space to reflect on their situation, a participant might become upset during an interview. The research team put a number of measures in place to deal with this possibility. (1) All interviewers were given training on how best to deal with such situations if they occurred, including guidance on the importance of a neutral, but compassionate, reaction, ensuring that the participant continues to feel in control of the interview and whether it continues, is paused or terminated; (2) in each gastroenterology clinics a named contact was identified who would act as a first point of contact should a participant express to the interviewer a need for support. Where a need for support was expressed, the participant or, at the participant's request, the interviewer got in touch with the named contact; (3) the LISA project research team put together a 'Useful Contacts' leaflet which provided information on national organisations and local groups (including Crohn's and Colitis UK) that offer support to people with IBD. Everyone who took part in the project was sent a copy of this leaflet on completion of their interview. (4) The lead researcher held regular debrief meetings with the project interviewers throughout the fieldwork period to address any concerns interviewers had about research participants, and decide whether any further action was required.

NHS Research Ethics and Research Governance Approval from each of the Research and Development (R&D) departments attached to the five NHS Trusts involved in the study was applied for and granted before the fieldwork began.

## **Data analysis**

All but one interview was audio-recorded (with permission) and fully transcribed. In the instance where it was not possible to record the interview, notes were made during the interview and written up in full once the interview was completed. In two cases, the interview was conducted in a South Asian language (one in Mirpuri and one in a mixture of Punjabi and English) and a back translation, focusing on conceptual rather than literal equivalence (Atkin and Chattoo, 2006), was carried out. The transcripts or summaries were then subject to a thematic analysis using 'Framework' (Lewis and Ritchie, 2003), a well-established method which supports the robust management of the data analysis process, including systems and processes which allow external scrutiny of the analytical process. This is a key requirement of rigorous, high quality qualitative data analysis (Pope et al., 2000). Key processes of thematic analysis are: data immersion; identification of themes; data reduction; data display; and analytical writing, including conclusion drawing and verifying (Miles and Huberman, 1994).

To assist with data immersion, the lead researcher (R1) carried out an initial read through of all transcripts and the field notes written by the interviewer, and then wrote a 'pen portrait' for each interview. A pen portrait is a 1-2 page summary which 'which makes the person come alive to the reader' (Holloway and Jefferson, 2000). It includes any inconsistences, contradictions and puzzles noticed within the data. In addition, a list of all themes observed in each interview was drawn up.

The process of moving from this initial list of themes to final thematic framework was lengthy and a full description of the steps taken is given in Appendix 3. In order to ensure rigour in the process, a number of systems were put in place. First, a teambased approach was taken, with three members of the team involved in developing the thematic framework. Second, a 'Review and Reflection' meeting was held for LISA Project interviewers, where they were presented with an overview on the themes emerging from the data and asked whether these themes fitted with their understanding and interpretation of the interview data. Third, the research team reported on the draft thematic framework at the second meeting with the PAG, and asked for their feedback on themes emerging, as well as the team's initial interpretation of the data.

The final framework consisted of five main themes: Living with IBD; access to support; perceptions and beliefs about the causes of IBD; visits to South Asian countries; feedback and recommendations to Crohn's and Colitis UK (see Appendix 4 for full list of themes and codes). Having finalised the thematic framework, R1 charted all transcripts, using Excel 2010 software to support the process. The 'chunks' of coded data were summarised, participant by participant, into the relevant chart. Each participant's data was entered into a different row. Small portions of verbatim text (or quotes) were also transferred onto the charts. Each chart entry was

referenced back to the transcript by noting the transcript page number alongside each summary or quote. Each chart thus displayed each individual's views/ experiences relating to a particular theme/issue, as well as allowing the accounts of different participants, or groups of participants, to be compared.

The charts were then explored and interrogated, with analytical notes being made of this process. Given the aims of our research, we were especially interested in exploring the nuanced influences of ethnicity and culture, rather than assuming one or the other 'essentialises' and dominates a person's experience of IBD (Atkin, 2009). In effect, we were concerned with understanding when 'ethnicity' and culture make a difference and when they do not; and to what extent there is a generic experience of IBD, irrespective of ethnicity. For every theme, the data was explored to see if the experiences were essentially the same for all participants, or whether there were differences according to gender, faith, country of birth, age. We were especially keen to avoid using ethnicity as an essentialising variable, which explained all aspects of a person's experience. We wanted to understand ethnicity within the broader context of a person's life. Gender, for example, might be more important in understanding a person's experience than ethnicity. In addition, interview transcripts were searched for any explicit references made by participants to South Asian identity, culture, faith, traditions or community. Where evidence was found, the data was highlighted. Summaries of charts were then written up and the final structure of the report decided, with different members of the research team responsible for interrogating and writing up different themes. Finally, towards the final stages of writing, all members of the research team reviewed the findings.

# The research participants

The final sample of research participants included 20 women and 13 men aged between 18 and 65 years of age. Twenty people identified themselves as being Indian/British Indian; nine as Pakistani/British Pakistani and four as Bangladeshi/British Bangladeshi. A large proportion (n=19) were born in the UK. Table 1 and 2 provide further details on the ethnic background, faith, gender and age of participants.

Table 1: Country of birth and ethnic background of research participants

	Number
Country of birth	
UK	19
India	4
Pakistan	4
Kenya	2
Uganda	2
Bangladesh	2
Ethnic group	
Indian/British Indian	19
Pakistani/British Pakistani	9
Bangladeshi/British Bangladeshi	4
British Punjabi	1

Response forms completed by the 33 research participants prior to taking part in the project indicated that 16 were Muslim, nine Hindu, seven Sikh and one person had 'no faith'. While interview data revealed that most participants practised this faith, six people (three Hindu, two Sikh and one Muslim) said that this faith was *not* an important part of their everyday life, describing themselves as not religious, only attending to religious events to pay respects to others, breaking religious observances (i.e. drinking alcohol) and/or being drawn to faiths other than the one they had ticked on the form.

Table 2: Breakdown of study sample by gender, age and religious affiliation

	Age	Muslim	Hindu	Sikh	Other	Sub-total
Male	18-24 yrs	2	0	0	0	2
	25-44 yrs	3	1	2	0	6
	45-65 yrs	2	1	2	0	5
Sub-total		7	2	4	0	13
	Age	Muslim	Hindu	Sikh	Other	Sub-total
Female	18-24 yrs	2	3	0	0	5
	25-44 yrs	5	3	2	0	10
	45-65 yrs	2	1	1	1	6
Sub-total		9	7	3	1	20
Grand total		16	9	7	1	33

Participants had been diagnosed with IBD for between three months and 21 years (median six years). Eighteen had CD and 14 had UC. Almost all had been prescribed steroids and over two-thirds had been prescribed immunosuppressants for their IBD at some point in time. Many had been also been prescribed biologics (n= 13) and/or been through surgery (n=8).

Table 3: Health background of research participants

	Number
Diagnosis	
CD	18
UC	14
Participant unclear	1
Surgery for IBD	
Yes	8
No	25
Ever prescribed steroids	
Yes	31
No	2
Ever prescribed immunosuppressants	
Yes	24
No	9
Ever prescribed biologics	
Yes	13
No	20

At the time of the interview, 17 participants were in paid employment (either full or part-time). Another six people described themselves as having caring responsibilities for either children or ill parents, five were students, and one was involved in voluntary work on a daily basis. Just one person described themselves as unemployed and another said they were on long term sick leave.

 Table 4:
 Socio-demographic background of research participants

	Number
Marital status	
Married	19
Single	12
Divorced	1
Separated	1
Employment status	
Working (full or part-time)	17
Caring responsibilities	6
Studying	5
Voluntary work	1
Unemployed	1
Retired	2
Long term sick leave	1

## **Chapter 4** Living with IBD: The Shared Experience

## Introduction

Interviewees typically struggled, at least initially, to identify ways in which their experiences might be different from the rest of the UK IBD population. For some this was because they were born and brought up in the UK and as a result had a reduced sense of South Asian identity and/or limited contact with the South Asian community, seeing themselves as 'British' or 'Westernised'. For others, it was because their experience of living with IBD was driven symptoms such as pain, tiredness, diarrhoea, and faecal incontinence and, as these having no ethnic or culture boundaries, it was hard to see how their experiences or needs might be different to others living with the condition:

'I don't think it makes much difference whether you're Sikh or Indian or English or anything. Pain is pain at the end of the day'.

[Female, Indian/British Indian, Sikh]

#### And

'I never really thought of it like that, like the South Asian woman type thing, I just think as a patient'.

[Female, Pakistani/British Pakistani, Muslim)

This view that there is a generic IBD experience is important because it reminds us that South Asian patients have a great deal in common with the rest of the IBD patient population and much of the support they need and want will be the same. In order that this is not forgotten, and so that later data is contextualised, in this chapter we report on experiences for which we could find no evidence of influence by South Asian culture or faith. As we shall see, there does seem to be a generic experience of IBD irrespective of ethnic or cultural origins.

## **Dealing with physical symptoms**

In this study participants varied considerably in the extent to which they experienced symptoms of IBD, from those who reported that their condition was well controlled and had little impact on their everyday life, to others who experienced frequent flare ups, spending periods of time completely debilitated by the condition, often housebound. However, 'flare ups' were not the only issue for participants. In between flare ups, many patients spoke of ongoing, milder but nonetheless troublesome symptoms (usually pain and fatigue), and worry that their health might worsen.

All participants spoke of the efforts they made to control or minimise the physical symptoms they experienced, and improve their sense of well-being. The most common approach was to seek medical advice and take up recommended treatments. In a number of cases people struggled with the side effects of medication, but for most people the benefits (e.g. avoiding pain, avoiding surgery, being able to work etc.) outweighed the disadvantages. Some spoke of how they had become more compliant over time, either through nurses explaining how the medication worked and would help them, or through experiencing how their health fluctuated according to the medication they were taking. Where patients did *not* take the medication prescribed, the main reason given was side effects. All concealed their non-compliance from clinic staff, though sometimes this was not successful. In addition to medical treatment, often driven by beliefs that IBD symptoms were exacerbated by particular foods and/or stress, people also spoke of: adjusting their diet; limiting the amount they ate and/or fasting; complementary and alternative treatments; lifestyle changes (e.g. slowing down, exercise); seeking the help of faith healers; adopting cognitive strategies (being positive or stoical), or turned to their faith and religious practice to help them cope.

## **Support from others**

Participants did not manage their IBD in isolation but were offered and accepted support from others. Just as might be expected for any group of IBD patients, there was variability in the support available, and who people turned to for help, depending on their life stage and personal circumstances.

Most young people living at home seemed to receive a great deal of practical support from parents (e.g. having meals prepared for them; being taken to hospital appointments; help with self-care when very unwell; and, in some cases, help with the administration of enemas), as well as emotional support, such as parents ensuring they 'kept their mind healthy' by continuing to go out during periods of illness. While many said that extended family (aunties, uncles, cousins and grandparents) knew of their health difficulties, few seemed to get a great deal of support from these people. Instead they spoke of these family members finding it difficult to accept that they had a long term illness, repeatedly asking when they would get better, and/or questioning the treatment they were receiving. Young people also spoke of talking to friends about having IBD, but were selective about who they talked to and what they told them, keeping it a small group of close friends and not divulging a great deal about symptoms. Very occasionally friends visited the young person in hospital, or offered practical help, such as collecting notes from lectures.

For most married participants their main source of support was their spouse and overall spouses were reported to be very supportive. Support was practical (i.e.

taking over cooking, cleaning and childcare tasks etc.), health related (i.e. administering enemas, managing medication regimes, gathering information on IBD and treatment options etc.) and emotional (i.e. encouraging and helping the person to go out, encouraging the person to put their health before other priorities, calming the person down when stressed/upset etc.). However, there were also arguments and moments of tension and, in a few cases, support was said to be limited because the partner was tired from working or the couple did not have a close relationship.

Married people varied in whether they told extended family and friends about having IBD. Often they told people about their illness when it became obvious they were unwell due to fluctuations in their weight or a hospital stay. Conversely the illness, or at least the extent of the illness, could be kept hidden from people who did not live nearby and with whom participants only had telephone contact. The amount of practical support offered to participants by extended family was limited. While a couple of participants spoke of siblings and parents living nearby and being around to help (e.g. accompanying the person to hospital appointments, cooking meals during periods of illness etc.), it was more common for participants to report they did *not* have family members who were in a position to offer support, making comments such as the following:

'Everybody has their own life. Everybody has children. My sister has five children. Nobody can really help. The best they do, is if they come to visit me, they make their own tea. You know in this country everybody is busy with their own children and work. The most support comes from my husband. When I'm ill he helps and does everything'.

[Female, Pakistani/British Pakistani, Muslim]

A very small proportion of participants were living alone. Those who were younger tended to be in regular contact with family and friends, and could turn to them for support when necessary. In contrast to this, older single participants had few family or friends who they saw on a regular basis. Out of the all research participants, this group seemed most alone in dealing with their illness.

## The emotional toll of IBD

Despite the efforts people were making to manage their condition, and the support received from others, the negative impact of having IBD on emotional well-being was a significant and frequently occurring theme in interviewees' accounts of their lives. It was apparent that many people were struggling to cope not only with the physical symptoms of IBD, but with the toll these symptoms had taken on their emotional well-being. Most commonly people spoke of experiencing periods when they felt 'low' or 'hopeless'. For example:

'You get depressed, you're trying to think of what I've done wrong, you know, in a sense, you know, why, cos your pain is such a severe pain, you can't even think of anything else'.

[Male, Pakistani/British Pakistani]

#### And

'It's just a battle everyday like. Sometimes I feel all right, but most of time like, I don't know, I feel quite down'.

[Female, Pakistani/British Pakistani, Muslim]

A very small minority spoke of times of intense despair and of wanting to 'give up'. One interviewee had attempted suicide on a number of occasions. Comments such as the following were made:

'You get to that stage you will say, what's the point of living, when, you know, sometimes it's so bad that you, you just don't want to be in that situation'.

[Female, Indian/British Indian, Sikh]

#### And

'I feel like I'm not living life, I'm existing. When I count the years, I'm going, "Okay, now you're probably going to live 'til you're 65" and I think, "You're 46 next year, how many years have I got?" That is it; that is my life. So I'm thinking how much longer have I got? It's almost like waiting for it to end, for life to end. I want to believe in reincarnation and maybe the second time round will be better'.

[Female, Indian/British Indian, No religion]

A number also spoke of feelings of anxiety and/or panic. Most often people described these feelings being brought on by worries about needing the toilet urgently, as described by the following participants:

'We've just left the house ten minutes ago, why couldn't you have gone then? I didn't need to go then, now I need to go .. and then I'm in a panic even more because I think oh I'm not going to get to the toilet on the journey there .. so that makes me want to go even more. It might not be a flare up, I think it's sort of probably psychological half the time'.

[Female, Indian/ British Indian, Hindu]

#### And

'I'm a bit wary now, of sort of going on a long trip, because with the problem I have got is I can't really hold my bowels too long, you know, just literally minutes, just a few minutes and I've got to get to a bathroom. It may not, you know, may not ever happen, I may not ever have an accident. But it's just this fear that, you know, if I'm say, you know, travelling on the motorway, it's always going to be nagging in my head, oh no, no, if something happens there's

nothing – you know, I'm not going to get to a bathroom for another 30 miles. So, it's just things like that really that...'

[Male, Indian/British Indian, Sikh]

Other worries, such as minor symptoms which might indicate the onset of a flare up; worries about eating the wrong foods and triggering symptoms, and thoughts about developing cancer were also mentioned.

Some people spoke of how their low moods, including feelings of irritability and anger, affected not just themselves but also the rest of the family, including both spouses and children:

'I try my family .. telling them "Do not say anything which I do not .. I dislike, or I don't like it, try to avoid it. And when I'm in a position where I, I'm not well, I don't like certain things, do not argue with me" because I get very angry and I shout at them'.

[Male, Pakistani/British Pakistani, Muslim]

#### And

'Sometimes I'll get agitated, you know, it'll affect my mood when I'm not well, because I'm so tired cos of the pain .. I find it affects my mood, and I, I get upset with that because then it's a knock-on effect on everyone else isn't it? And if you're kind of the backbone of the family, so if it affects you, it's gonna affect, it's a ripple effect, it's gonna affect everyone else isn't it?'

[Female, Indian/ British Indian, Hindu]

## Impact of IBD on everyday life

Although participants were making considerable efforts to manage their IBD, nevertheless it remained the case that IBD impacted on many aspects of participant's lives, with some variation according to life stage.

#### Social and leisure activities

An issue for participants of all ages and life-stages was the impact of IBD on social and leisure activities. People spoke of difficulties taking part in family activities or outings (e.g. going to the park, cinema, shopping, or to visit friends and family etc.), avoiding meeting up with friends and family, or attending social events, as well as giving up taking part in or going to watch previously enjoyed sport. In some cases, people's social life was only affected when they were having a flare up. For others the impact was more pervasive, with them explaining that even when they were *not* having a flare up they avoided going out socially because of milder ongoing symptoms such as pain and/or tiredness, fears that they *might* experience

symptoms, worries about whether it would be possible to get access to a toilet when out, embarrassment about using toilets in public places or other people's homes, and generally feeling unhappy or 'low':

'I don't actually socialise, I don't go out, I don't feel comfortable going out with the condition I've got .. because I feel too tired.....I just don't make plans because I don't know what the severity of the pain is going to be like. I just don't make plans...because the severity of the pain varies day to day'.

[Female, Indian/British Indian, Sikh]

#### And

'I think with the social gathering bit I would just, even if it wasn't a day that I was feeling particularly unwell I'd probably avoid, or maybe avoid it .. because I don't know, it, it feels like you've got so much going on and then everyone's just going to be talking about their hairstyle or their nails or something and, and you're not really bothered, so I'd just avoid going'.

[Female, Indian/British Indian, Hindu]

Some young people also spoke of not even having the energy to telephone or text friends. In a few cases people spoke of gradually losing touch with friends after seeing less of them, making it difficult to re-establish a social life when they were feeling well enough to do so:

'I used to go out drinking with friends, but I can't do that so much now because of the colitis. It's spoiled my life in some ways, you know, my interactions with other people, playing golf at the weekend; I don't go out so much now. I used to play golf at the weekends, and I really enjoyed that. I had lots of friends there, my golfing friends, and we would have a drink and a chat afterwards; that was very nice. But you know, there are no toilets on the golf course of course [laughs] and so it became too difficult, and eventually I stopped going there. So it's very sad, and I don't see those friends now, not really'.

[Male, Pakistani/British Pakistani, Muslim]

## **Education, employment and finances**

Younger people spoke of struggling to complete further education (HNDs and degrees), particularly when the course involved placements. Many managed to complete their course, though it took more time than it should have, but a few did not. A couple of young people who were ill in the period immediately after completing a degree or training course said that, even though they had now recovered, they were struggling to find their first job.

Amongst participants who had experience of employment, some spoke of IBD causing some to give up work completely, a few to work part-time, one to take early retirement, and another to be on long term sick leave. Others spoke of how they managed to work by: changing to a less stressful job or one that involved less travel; trying not to work excessive hours; using annual leave to rest at home during a flare up; and not eating or drinking prior to work so as to avoid an upset stomach/bowel.

Most participants chose to tell colleagues or employers they had IBD, explaining that people would find out anyway due to the need to take medication at work or have time off for treatments and investigations. Responses had been mixed. There were examples of extremely supportive work colleagues and employers, with people speaking appreciatively of colleagues who visited them when they were ill, and employers who had offered flexi-time contracts and offered to pay for private medical treatment. A small number of people spoke of having a very difficult time, with employers placing increasing demands on people despite knowing about ill health, or threatening to take disciplinary action due to lateness for work when unwell. A few people who were unable to secure flexi-time or short working hours spoke of struggling with work and worrying whether they would be able to continue in their current job.

A number of people spoke of financial difficulties, explaining that they had run into problems for a number of reasons, including: the need to work part-time; the need to change to a less demanding but lower paid job; spouses reducing their working hours because the interviewee being too ill to look after children; being refused disability living allowance; and/or no longer being eligible for sick pay.

## Relationships with partners and spouses

Although most people who were married had very supportive partners, a number spoke of how having IBD had at times put a strain on their relationship. This included both major and everyday sources of stress or strain. For example, financial worries caused by having IBD, the strain caused by a partner having to take over the running of the household, being too ill to be interested in sex, and the inconvenience of the frequent need to find a toilet when out. Two people who were single spoke of how having IBD had contributed to the breakdown of their relationship with previous partners, both attributing this to not being well enough to go out and socialise. The period prior to or shortly after diagnosis, when partners had limited understanding of the interviewee's illness was reported to be particularly difficult. For example:

'At the beginning when it was .. it was all really bad .. I, it was really hard in the bedroom just to be sexually active, it was so hard. .. I couldn't make him understand that, you know, I was having stomach pain, stomach pain, we were arguing constantly because he'd think I've had something bad to eat or, when

he's at work, or having too many chips or too much curry or what not. So it was really hard to make him understand, and things were really emotional for me back then because I, I didn't know what was wrong with me to make him understand, so it was hard for him cos he, he's had to basically cut down his hours at work .. just so he can be at home to look after our daughter, cos she was only one year old then. So it was very hard at that time for me. I mean there was a point was where things were stretching so hard that I just felt like I wanted to run away from this relationship. It was really bad'.

[Participant's characteristics withheld to ensure anonymity]

## Parenting and caring responsibilities

Parents, mostly but not exclusively mothers, spoke of what a struggle it was to care for young children. Pain, tiredness and frequent trips to the toilet made tasks such as taking children to and from school, cooking, washing and dressing children, keeping up with physically active children, and helping with homework difficult:

'They all need their hair done. Bathing them and getting them dressed. They are all young and need their mother, I'm their mother. They are not mature enough to take care of themselves. When it comes to children, you have to take care of everything for them. You need to make sure they eat. A mother will make dinner for her children. But when the mother is ill who will take care of them?

Interviewer: How does the mother cope?

Participant: With great difficulty'.

[Female, Pakistani/British Pakistani, Muslim]

#### And

'I don't want to go out, I want to rest quietly on my own and sometimes it's difficult because I don't want to play with the kids, you know, staying upstairs in bed; I don't go out, I don't go to the park with kids, don't go out, do the shopping, so it's very hard'.

[Male, Indian/British Indian, Hindu]

Mornings were particularly problematic because of frequent and urgent need for the toilet at this time.

# Chapter 5 How Does South Asian Identity, Culture and Faith Influence the Experience of Living with IBD?

## Introduction

This chapter reports on how South Asian culture and faith influence people's experience of living with IBD. As we have seen many of participants' experiences were generic and might be experienced by anybody with IBD, irrespective of their ethnic background. However, as interviews unfolded many described how aspects of their South Asian identity (i.e. faith, language, ethnic background, country of birth) and relationship with others in the South Asian community affected their experience. Analysis revealed that many of the issues were concerned about the social context in which people lived with and managed their condition, with a number of cultural factors influencing both how family, friends and members of their wider community responded to them having IBD and, in turn, how participants felt about having IBD and how best to manage it. These cultural influences were numerous and wideranging, including:

- culturally specific difficulties understanding IBD;
- the stigma of ill health within South Asian communities;
- the perception that bowel movements are a taboo subject;
- beliefs about the links between food and health;
- alternatives to conventional medical treatment;
- cultural expectations about a woman and a man's role with the family;
- 'Spicy' food is the norm;
- living with extended family.

In addition, participants described challenges which we would argue are more likely to be encountered by, but are not exclusive to, people who are South Asian. These were:

- making visits to extended family living overseas;
- practising faith.

These cultural influences and additional challenges are described further below.

## Culturally specific difficulties understanding IBD

A number of people spoke of it being difficult for family to understand IBD because it was not a condition they had heard of before. Many also spoke of not knowing

anyone from their own community who had IBD and/or that it was rare in the country their family originated from:

'If I say to someone who's non-Asian 'I've got Crohn's', they'll say, 'oh yeah, my brother's got that', or, 'my boss at work' or, you know, someone. If I say to someone who's Asian like I've got Crohn's, they wouldn't have a clue what it is probably, or probably they do know someone that has it, but cos they don't talk about it they don't know about it'

[Male, Indian/British Indian, Hindu]

Language barriers could make it difficult to explain the condition to family, with some people mentioning that the situation was made worse for them due to there being no word for Crohn's in Punjabi:

'Of course they have no understanding about the Crohn's, even abroad as well, they, all they know is about TB or... they have infection, or ulcers, but Crohn's is something unfamiliar abroad. What is Crohn's? But this is not a name abroad, in the UK or any foreign country, they have given it Crohn's or colitis, but abroad is, this, this thing is totally unfamiliar, they have not given this name'.

[Male, Pakistani/British, Muslim].

The term 'inflammatory bowel disease' also caused some confusion for some participants' families since 'disease' had different connotations in South Asian countries, the implication being from a number of participants that a disease was generally considered infectious and/or life-threatening:

'The first day I told my husband he was gobsmacked because he's never heard of having a disease where you don't end up dying'.

[Female, Bangladeshi/British Bangladeshi, Muslim]

A perception that others, even close family members, would not understand the condition prevented individuals telling people about their condition. This seemed to apply particularly to older family members and those not living in the UK:

'I haven't told my brothers and sisters or anyone, I just feel like it's embarrassing for me and I just feel like, you know, to tell someone that you, you've got a disease .. just like the reaction I had at the beginning, oh it's something that's catchy, it's something that you may end up dying, I don't want no-one to feel like that, I don't want people to start treating me differently... Nobody else apart from husband and father knows - to be honest, not even my in-laws abroad..., they're not all that highly educated, they're just going to take things the wrong way round'.

[Female, Bangladeshi/ British Bangladeshi, Muslim]

#### And

'I just feel embarrassed to tell anyone really, especially my mum. If I tell her then, I don't know, just talking about it out of context, she's more on the traditional side, she'll think something's come over me like, I don't know, like, you know, black magic and all that stuff. She's more into that, she'll say, oh something's come over you, or like, let's make you see a Moulana or all that and everything. I just don't want to go through that with my mum'.

[Female, Bangladeshi/British Bangladeshi, Muslim]

In a small number of interviewees (all Muslim men), and their families there seemed to be a culture of self-blame for illness, sometimes connected with some form of religious transgression. Muslim men who had moved to the UK from Pakistan prior to diagnosis, as well as their family, blamed their ill health on drinking alcohol and/or eating 'junk food' when they moved to the UK. For example:

'People say it's cos I've been eating most of the time outside [of the home] like. That's what I was thinking that's why it happened and maybe also that I've been eating outside food, takeaway food'.

[Male, Pakistani, Muslim]

Another spoke of his parents believing that him having IBD was a punishment for behaviour which had led to him spending some time in prison. In one case, a participant spoke of how perhaps if he had been a better Muslim, and not been so greedy, he would not have developed the condition. Here beliefs about what it was to be a good Muslim were combined with beliefs about the impact of lifestyle on IBD, with him explaining that because he had been greedy, he had worked too hard, and that this lifestyle had brought about ill health. One participant expressed the view that IBD might be sexually transmitted and that the increase in prevalence in South Asian patients was due to people having relationships with someone who is White British.

## The stigma of ill health within South Asian communities

Some participants said that they did not tell others about having IBD because within their community being ill reflected badly upon people, impacting on prospects for an arranged marriage and raising concerns about the health of their children or any children they might have. For example:

'So I think, yeah, people tend not to talk about, you know not only IBD but like if they have any illness in general.... the Asian community like arranged marriage is still prevalent, so if you were, wanted to do that and you said that, that you've got this then .. people, like an opposite party is less likely .. going to want to marry you...I think they also look for like .. they don't want it to seem like you've got bad genes or anything cos then you could like pass it down to your children or whatever'.

[Male, Indian/British Indian, Hindu]

#### And

'If you want an arranged marriage, if you are gonna have an arranged marriage, it probably is hard to find somebody because everybody will always think you're just ill, and nobody, I think in, in Asian culture everybody wants this perfect partner who's, you know, got everything and, you know, illness plays a big role, nobody really wants to be with somebody; it sounds really awful but there is, you know, this cultural expectation that you're fit and healthy and they don't, especially for a girl I think, because they expect you to have this like housework, take on this housework role, and if you're not well then you're not gonna be able to look after the family and your husband and what not'.

[Female, Pakistani/British Pakistani, Muslim]

This view that having IBD could impact on an individual's marriage prospects was also shared by an older male participant, who felt sympathy for younger patients with IBD:

'It's .. you get marked, don't you? When, when you live in a close knit community .. say for example .. if you're a young girl or, unmarried young girl or a man and .. people find out you're suffering from this condition .. I know for a fact that .. you know, people will look at, twice at them, you know, for relationships and stuff like that.....and .. other issues .. as well. So .. some people might even think it's contagious or something like that'.

[Male, Indian/British Indian, Muslim]

In addition, there was a perception that rumours about your health will spread quickly throughout the South Asian community, resulting in a loss of privacy and unwelcome visitors when you are in hospital. The following extract illustrates this issue:

'I think being, Asian people aren't, everything's taboo, everything's hidden, don't tell anyone and that's just the way things are. I think that's changing now as time's going on, people are talking and discussing. if you had a certain condition you'd end up, you know, through Chinese whispers, as having a different condition or being worse than it actually is, or somebody will say other things. So people tend to not say things as much. Initially that was the reasons, you know, we don't want to tell everyone cos people will start changing things and making things worse than they actually are, or people will come hounding in at, at the hospital, we don't like anybody at the hospital, we just want to be left alone .. that was the reasons why I didn't tell anyone .. and it's just sort of

carried on that you don't want to be labelled as different or weird or got a condition or a, .. something wrong with you'.

[Female, Hindu, Indian/British Indian]

## The perception that bowel movements are a taboo subject

A number of people expressed the view that IBD is a taboo subject within South Asian communities, particularly within the older generation. This meant it could be difficult to discuss health problems with family members and/or members of the wider community. In discussing this, participants acknowledged that talking about bowel related issues can be difficult for anyone, whatever their background, but maintained that these difficulties were greater for people from a South Asian background, comparing their experiences talking to people from different backgrounds as evidence of this. For example:

'I had a colonoscopy there [Pakistan] and it was, it was awful, like they just, it's just not something they speak about. Or even when I went to see the doctor I just did not feel comfortable telling him about my bowel movements, whereas when I go to the doctors here [UK] it's fine. But I really did not feel comfortable at all discussing my bowel movements there, with him or actually the other, any other Asians really, whereas over here it's, it's more acceptable I think to discuss'.

[Female, Pakistani/British Pakistani, Muslim]

#### Beliefs about the links between food and health

Interviewee's reported difficulty getting others to accept their IBD related dietary restrictions because of beliefs about food. First, some spoke of there being a perception that if you change your diet you will no longer be ill because of beliefs that health problems are the result of an imbalance in the foods that you eat. Second, when a food is considered blessed within a particular faith, it can be difficult to get others to understand or accept an inability to tolerate it. As one man explained:

'I think the biggest problem will be some people will not be able to differentiate from their religion to medicine. They just can't seem to see why this food is breathtakingly bad for me'.

[Male, Indian/British Indian, Sikh]

## Alternatives to conventional medical treatment

Many participants spoke of using some form of complementary or alternative medicines or treatment (CAMs) to help them manage their condition, including: Aloe Vera gel; wheat grass; fennel seeds; Isabgol<sup>3</sup>; ayurvedic medicine<sup>4</sup>; herbal oils and tablets; homeopathic medicine; Reiki healing; hypnotherapy; yoga; and reflexology. Reasons given for using such treatments were: that people would rather do something that helps alleviate symptoms a little than do nothing at all; that it helps give you a more positive mental attitude to your condition; wanting to feel that something healthy was going back into their body (to counteract the effects of medications); wanting to do something that might reenergise their body; and feeling that giving themselves some 'TLC' (tender loving care) helped to make them feel better. In most instances, people said they did not know what was in the homeopathic or herbal remedies taken. It was notable that a number of people felt such treatments would not do any harm as they were 'natural' or herbal'. However, a few had suffered side effects and were surprised to learn from their IBD specialist nurse that some herbal remedies do not mix well with IBD medications.

Although anybody with IBD might choose to use such treatments, the data suggested that South Asian patients may be more likely to encounter such treatments during the course of their everyday life. It was notable that some of the CAMs mentioned were traditional Indian medicines or remedies (e.g. ayurvedic medicine, Isabgol) or participants mentioned buying them from 'Indian shops'. One person spoke of how in certain cities in the UK with a high South Asian population there are many shops selling herbal remedies. A few people spoke of seeking out homeopathic treatment when they had visited family in South Asian countries and one man spoke of how he planned to look into herbal treatments the next time he made a trip to Bangladesh as they were widely available and used to treat a number of illnesses. A number of patients also spoke of family members, encouraging them to seek out/or use CAMs, including 'Asian remedies', either within the UK or abroad.

\_

<sup>&</sup>lt;sup>3</sup> **Isabgol** (also known as Psyllium seed husks) are portions of the seeds of the plant *Plantago ovata*, a native of India and Pakistan. Psyllium seed husk are indigestible and are a source of soluble dietary fibre. They are used to relieve constipation, irritable bowel syndrome, and diarrhoea. They are also used as a regular dietary supplement to improve and maintain regular GI transit. The husks are used whole in their natural state, or dried and chopped or powdered for easier consumption. In either of these forms, one takes them by mixing them with water or another fluid.

<sup>&</sup>lt;sup>4</sup> **Ayurvedic medicine** is a system of Indian traditional medicine which originates from the Indian subcontinent and is based on the belief that health and wellness depend on a delicate balance between the mind, body, and spirit. The primary focus of Ayurvedic medicine is to promote good health, rather than fight disease, but treatments may be recommended for specific health problems. Ayurveda is at present well integrated into the Indian National health care system, with state hospitals for Ayurveda established across the country (WH0, 2014) However, concerns have been raised about Ayurvedic products; with some U.S. studies have shown that up to 20% of Ayurvedic U.S. and Indianmanufactured patent medicines sold through internet contained toxic levels of heavy metals such as lead, mercury and arsenic (Saper et al, 2008).

However, it should not be assumed that all South Asian patients will be encouraged to use, or will automatically take up, such treatments. Some participants had family members who were health professionals and who had warned them against the use of CAMs which were not tried and tested. A number of participants spoke of their concern about taking any form of alternative remedy, sometimes because they had witnessed a family member use CAMs unsuccessfully, and expressed a preference for medical treatments. For example:

'I just feel like the doctors are in their places for a reason, the consultants are there in their places for a reason, we should just leave these things up to the professionals'.

[Female, Bangladeshi/British Bangladeshi, Muslim]

Furthermore, although a couple of interviewees referred to pressure from one or more family members to seek help from a faith healer, this was resisted on both occasions due to feeling that it would not help, believing that the faith healer might be a 'Charlatan', and/ or feeling that seeking help from such people was not part of the Islamic faith. Just a small number of people said they had been to see some form of faith healer, and in two instances these had been 'English healers' and not someone from their own faith. One Muslim had sought help from an Imam and had received prayers and holy water. In all of these cases, the help of a faith healer was being used in addition to, not instead of, medical treatment.

## Cultural expectations regarding a woman's role within the family

A number of female participants spoke of how having IBD impacted, or could in the future, impact their ability to be a homemaker and/or mother. This concern was not just about the practical difficulties of managing these roles and responsibilities when ill, but also about the social consequences if they failed to do so, with interviewee's talking about how people within their community would think badly of them, with one person going as far as to say that people might think she has been affected by black magic if she is seen to be unable to look after her home:

'I do understand once I do have a family of my own it'll be much more difficult. If I don't have the energy to do things like that now I can't imagine how it will be then. So it does, that is a bit worrying.....I think that can be quite difficult for Asians to understand, South Asians. I think the fatigue thing as well cos people just think you're really lazy and there's like these expectations that you should be doing all the housework and things, because it's just like Asian culture, you know, you're expected to be like, like a housewife and if you don't do all that they just think you're useless and lazy'.

[Female, Pakistani/British Pakistani, Muslim]

And

'I mean if people, my family or anyone find out that my husband's helping with cooking and cleaning, not everyone's going to be happy about it, everyone's going to say well what's the wife there for? The husband has to go to work, has to cook, has to clean, get the children from school and everything, what's all that about, you know, what's the wife doing? It's really hard to explain to people that when you're unwell you're unwell, you know, you call it a day you can't do anything, people just expect, you know, wives get, probably get just the old sniff or flu or cold and, but they still get on with it, they still make the chapattis, they still make the rice and stuff, but with Crohn's it's totally different, it's totally the opposite, you can't do anything'.

[Female, Bangladeshi/British Bangladeshi, Muslim]

A couple of recently married women spoke of how their new role as wife obliged them to attend family gathering and celebrations in a way that had not been expected prior to marriage:

'Before when I was living at home I was able to say to my mum that I don't think I can go to this wedding or I don't think I can go, you know, to someone's party or whatever. Now it's not like that. It's almost like when you're a daughter-in-law you're expected to go to everything'.

[Female, Indian/British Indian, Hindu]

Negotiating non-attendance of such events when unwell was made more difficult for these participants when they had not told others outside the family that they had IBD.

Sometimes concerns were specifically about difficulty becoming pregnant due to having IBD:

'If I'm honest, there's just that big burden in your head that, you know, I get the pressure in our culture just to say why aren't I getting pregnant again, that's, that's one of the big pressures that I get. My sister, she's got two kids, and then my younger sister, she's expecting, you know, people just question the fact "oh why aren't you getting pregnant, what's wrong with you?" etc. etc. but you can't go around telling everybody I've got Crohn's disease, I'm unwell, do you know, that's why I'm not getting pregnant maybe as quick as a normal person will'.

[Female, Bangladeshi/British Bangladeshi, Muslim]

#### And

'Really I was supposed to be married and have ten kids by now, the Asian way. But yeah, life has just taken its own turn and I'm just taking it day-by-day and just kind of going with what I've got now. .. I'm gonna have IVF and have a baby, so I'm, and that's my hope, to have a baby, even if I just have one'.

[Female, British Punjabi, Hindu]

## Cultural expectations regarding a man's role

It is important to acknowledge that men also felt pressure to meet culturally prescribed obligations, with one man describing how his ill health meant it was a struggle for him to be the father figure he was expected to be to his brothers, providing financial help and guidance, since his father passed away. He also spoke of his difficulty accepting that his wife needed to go out to work because of his ill health:

'Sometime I don't feel it's right thing to do it, you know, for a woman, even though it's, Asian family it's just a tradition, you know like men should be working, women stay .. but it's just a modern environment, so of course, she used to work, she used to work before she was born [their child]. But now she saying "OK, if you take a rest and let me just take over and, you know .. carry on .. work, you know, and you just take a rest" because the situation's deteriorating a little bit'.

[Male, Pakistani/British Pakistani, Muslim]

## 'Spicy' food is the norm

Most participants lived in households where eating traditional South Asian food was the norm. This was a significant issue for participants since the majority of participants believed that 'spicy' food, specifically food that contained chilli, was something they needed to avoid. Problems were also reported with vegetarian food, which often featured in family meals. Almost all participants who felt there was a link between South Asian diet and IBD felt that it exacerbated, rather than caused, their condition, though there were exceptions. For example:

'We have a lot of salt and chilli and dairy milk in our diets and that could again maybe contribute towards the Crohn's ...and maybe also the chilli. If you put chilli on your skin, if you ever put it on the skin and you're keeping it on the skin, eventually you're going to get – you're probably going to get – it's going to go all red, isn't it? Start stinging, etc. So if you're ingesting it...every day it's in the curries and it's got chilli, always got chilli every day constantly, week in week out, a couple of times a week, that chilli is going to start burning your insides, yeah, no? I don't know and this is just a thought maybe'.

[Female, Indian/British/Indian, no religion]

Where the person had responsibility for cooking, the need to avoid such foods was manageable, with participants describing how they modified meals and how, in general, their family had learnt to accept these adaptations. However, where the person was not in charge of cooking (usually a young person still living at home with their parents or a married woman living with her in laws) this could be difficult as

sometimes family members did not understand or accept the person's inability to tolerate spicy and/or vegetarian food. As a result some people spoke of eating food they knew would make them ill. For example:

'It was a struggle trying to get somebody to understand what non-spicy was, cos it was just wasn't, cos they used to think it was not spicy. Our definitions are totally different. .. But yeah, I think you just got on with it .. you just eat what's there, I'm lucky to have food on my plate, so yeah, it's fine, just eat it .. so yeah'.

[Female, Indian/British/Indian, Hindu]

#### And

'Or even at home if my mum's made something and I find it, if I tell her I, I find it quite spicy, she just thinks that, you know, it's just me being funny because everybody else is fine with it, and then because of that sometimes you just, you know, you try not to mention anything and you just carry on eating or whatever, which can affect you quite a bit, cos like I said I do think diet plays quite an important role in my IBD so it can flare up my Crohn's'.

[Pakistani/British Pakistani, Muslim]

South Asian diet also caused people difficulty outside the home. For anyone suffering from IBD, going to social events such as weddings or parties is difficult when unwell and experiencing fatigue, pain and or toilet urgency. For South Asian patients there is the added complication that attending such an event involves accepting food which is spicy and which they would rather avoid. Participants dealt with this in different ways. Some simply avoided going to family events, unless it involved close relatives. Others attended events but stayed a short time in order to avoid eating. Some attended social events and ate despite knowing they would be ill the next day. Visiting family members who were aware the person had IBD was less problematic, as under these circumstances people felt able to take their own food with them.

## Living with extended family

A number of participants lived with extended family and most described these family members as broadly supportive. However, women who had moved in with in-laws on getting married described how they initially felt embarrassed about symptoms and their frequent need to use the toilet. This caused one woman to spend much of her time when unwell in her bedroom in order to maintain some privacy. Most people said that over time, as family members became more aware of their health difficulties, embarrassment gradually disappeared. In a few cases participants also spoke of having, or feeling they had to, cook and clean for the whole family despite being in very poor health. Participants also pointed out that living with larger extended families could mean speedy access to a toilet was more difficult.

## Visits to extended family living overseas

A large proportion of participants spoke of making trips to South Asian countries, usually to visit family or attend a family event, such as a wedding. For many it was the family related nature of the trip that encouraged people to go despite a number of worries and concerns. The most common worry was about making the journey itself, with people referring to fears about being able to get to a toilet whilst on an aeroplane or when travelling long distances within the country:

'And my worry would be what if I don't get a toilet from Bombay just to where I'm going which is five hours' car journey... what happens if I can't get to a toilet or there's not a decent toilet to go to, or I have to keep stopping. I've already started thinking about that'.

[Female, Indian/British/Indian, Hindu]

#### And

'I've not been on a plane since 2007 because on a plane, you have to queue for the toilets, and you feel embarrassed if something happens to you, that's the reason'.

[Male, British Pakistani, Muslim]

Other concerns about making visits to South Asian countries were: worries about having to seek out medical treatment from people who are not familiar with IBD; getting a 'tummy bug' as this could have a huge impact on the health of someone with IBD; and difficulty disposing of stoma pouches. As a result of such worries some people had chosen not to make trips to South Asian countries. Where participants had relatives they were close to living in these countries, this could be a source of considerable upset.

Where people had made trips, they spoke of preparing for and managing such trips by: eating nothing at all whilst travelling; carefully planning of toilet stops (including seeking out mosques for their toilet facilities); and carrying medication and/ or their own food with them wherever they went. One patient went as far as to take a mini fridge on her trip in order to store humira injections at the appropriate temperature. Unfortunately, the medication became inactive due to it not being stored correctly whilst travelling. This in turn brought about a flare up in her condition, resulting in her having to cut short her trip in order to return to the UK for treatment. The person in question spoke of not fully understanding how quickly the medication became inactive when stored incorrectly.

A few participants spoke of being encouraged by family and friends to seek a second opinion or treatment for their IBD from doctors in South Asian countries and in a few cases people had done so. Others had sought help due to a flare up will abroad.

One spoke of running into difficulties because the doctor had not heard of ulcerative colitis and of how taking along leaflets in Punjabi to overcome any language barriers did not help:

'I took the Crohn's and Colitis .. Punjabi leaflets with me .. they haven't really heard much about it. .....and just to kind of explain it to Asians is really hard. They're like "What is this? What do you mean? Ulcers, where? What happens to you?" They're, they're just like oblivious like that. So that was a bit odd. ..'

[Female, British Punjabi, Hindu]

Another said that, although he had consulted a doctor abroad, he did not trust the advice offered so he had not taken the medication recommended:

'I don't trust that much the foreign medication, even though I'm bred and born there, you know, spend half my life there'.

[Male, Pakistani/British Pakistani, Muslim]

Perhaps because of the difficulties they anticipated, seven participants were surprised to find that their health improved while they were abroad:

'That was remarkable because I had no flare up when I was in Pakistan. I mean I thought I might struggle there, because there's a bit of a culture shock as well, you know, different foods, different environment, and I was staying there for four weeks .. I came out completely unscathed, it was a bit of a miracle for me, you know. I thought to myself I've had four flare ups already, now I'm going to Pakistan, I was feeling a bit worried about it, should I go, shouldn't I go, you know, with the different climate there, but I came through perfectly fine'.

[Male, Pakistani/ British Pakistani, Muslim]

People speculated as to why their health improved, most commonly referring to the warmer climate. Other possible explanations given were that it was less stressful abroad as they were looked after and/or well fed by family. Unfortunately, a few people did become ill while making trips. In most cases, people were clear as to why this had happened, referring to how they 'relaxed' and had stopped taking prescribed medication and eaten food they cannot tolerate.

## **Practising faith**

People of all religious denominations spoke of how their faith helped them to cope with having IBD, particularly at times when they were in pain. A number described: the calming, almost meditative, effect of prayer; the benefit of being active within their faith community since this provided a reason to overcome difficulties caused by

IBD. Muslim interviewee's in particular expressed the view that having IBD was God or Allah's 'will'. They explained that this helped them to view IBD not as a problem, but as a challenge sent by God, which they needed to both accept and rise up to. One participant took comfort from the thought that if she suffered in this life she would suffer less in the afterlife. A few people commented on the fact that they had learnt about the benefits of fasting as means of reducing IBD symptoms after fasting for Ramadan. Much of what patients had to say about their use of faith to cope with IBD is encapsulated in the following quote:

'It would be easy to say 'why me?' but being Muslim everything is seen as a test. They question is "How are you going to deal with it?" Even if you prick your finger on a thorn. The prophet was closer to God when ill and so I may be too. This is my destiny. If I didn't have Crohn's I wouldn't be as understanding of people as I am. Also, this life is temporary, which helps. What doesn't kill me makes me stronger. Also prayer helps to calm me down. This hospital Imam was also very helpful. He would try to give me a pep talk. Help me to see things in a positive way. Despite what they say in the media, Islam can be a positive thing and it's very relaxed. During Ramadan you don't have to fast if you are unwell, even if it's just a headache'.

[Female, Pakistani/British Pakistani, Muslim]

However, a common theme among Muslim interviewees was the difficulties they encountered in practising their faith. There were three main reasons for this. First, the act of Salat<sup>5</sup> is a physically demanding one, so was difficult for people to perform when in pain and fatigued. Secondly, participants explained that prior to praying they had to perform Wudu <sup>6</sup> and, if they were going through a flare up, this could be physically demanding as they would be required to repeat it prior to each prayer. Finally, people also spoke of worries about passing wind if going to the Mosque for prayers. As prayer was an important part of daily life, some pushed themselves to pray as much as possible despite these difficulties since not doing so led to feelings of guilt. Others prayed more when they were well to make up for missed prayer.

In addition to the difficulties Muslim patients had practising their faith, they also explained that the cleaning rituals required as part of their faith made it difficult to go out of the home when having bowel related symptoms since it was hard to find bathrooms with the appropriate washing facilities:

<sup>5</sup> **Salat** is the practice of ritualistic prayer in Islam. It is usually performed five times a day and preceded by ritual ablution. It consists of the repetition of a unit called a*rak* ah (pl. *raka* at) consisting

of prescribed actions and words.

<sup>&</sup>lt;sup>6</sup> **Wudu** - Before conducting salat, a Muslim has to perform a ritual ablution. Wudu is performed using water (*wudu*) and involves washing the hands, mouth, nose, arms, face, hair (often washing the hair is merely drawing the already wet hands from the fringe to the nape of the neck), ears, and feet three times each in that order. It is not obligatory to wash the hair three times, once is sufficient, and men must also wash their beard and moustache when washing the face.

'When you [someone who is not South Asian] would have to go to the bathroom, you would just go to the bathroom. When I'm outside [the home] I have to go to the bathroom where there's like maybe water facilities or I'll take a bottle or something with me. We clean ourself in a different way, because of our religion, so that would be a big issue for me. When I first got Crohn', and I was suffering from diarrhoea, it put a lot of fear on me. .. So yeah, it would probably prevent me going out for long periods of time'.

[Female, Indian/British Indian, Muslim]

## **Chapter 6** Experiences with Health Services

## Introduction

In this chapter we report on support from the two main providers of health services and support to IBD patients: General Practitioners (GPs) and gastroenterology services. Much of what people had to say was about their generic support needs, irrespective of their South Asian background. However, we report it here since we consider it important that this evidence is not lost, but is added to the existing literature on the support that IBD patients' value. Views on how well services attend to the needs of South Asian patients are reported later in the chapter. Finally, we end by reporting on the additional support participants would like to see provided by health services, including South Asian specific recommendations.

## **Support from GP services**

Very few of the participants in this study consulted their GP for treatment or advice on IBD. A wide range of reasons were given, including: feeling that their GP did not have sufficient knowledge to treat the condition; the gastroenterology clinic is quicker to respond to requests; the GP is not up to date on their test results and health status; prior experience of their GP being unsympathetic; and lack of continuity of care within general practice, with people seeing a different doctor every time they made an appointment. The following quote encapsulates many of these feelings:

'The doctors [GPs] don't give you any time, you only have ten minutes with them, and they don't listen to you. You see a different doctor every time, and you have to explain it all again, so they don't know what's going on with you – they are not interested'.

[Male, Indian/British Indian/Hindu]

In some cases, a breakdown in the relationship with the GP seemed to stem from the initial diagnosis, with participants stating that they felt disappointed with or let down by their GP during this time. Sometimes this was because the GP had failed to refer them for investigations despite numerous requests for help and/ or obvious signs that investigation was required (bleeding or extreme weight loss). Participants also spoke of the consequences of GPs being slow to refer them for investigations and treatment, saying this had led to: traumatic emergency admissions to hospital; severe pain; the unnecessary development of infections; lengthy hospital stays; long term consequences for their health; and, in one instance, the near loss of a baby during pregnancy.

Given these experiences it is perhaps not surprising that most participants chose to turn to gastroenterology services for IBD treatment and advice, and the remainder of this chapter focuses on these experiences.

## Support from gastroenterology services

When asked about support from gastroenterology services, it was clear that there was a great deal of good practice, with half the participants having only positive things to say about their experiences. Most of the remaining participants reported a mixture of things they liked and didn't like about the support received.

## Positive aspects of support provided by gastroenterology services

By far the most common reason for being positive about support from gastroenterology services was ease of access, with the majority of respondents mentioning that if they had any questions or worries they could get in touch with their specialist nurse or a consultant either by telephone or email, and that they would usually receive a swift response:

'My nurse gave me her phone number and email so I can call her if I want to, which is helpful. I called her once and explained to her my problems and she was very helpful, very nice. It's an excellent service I think, the clinic is excellent'.

[Male, Pakistani/British Pakistani, Muslim]

#### And

'I've got to say she's probably one of the best .. you know, nurses I've come across in my life. .. From the minute I was introduced to her .. up until even today, she's still with us, she's only a phone call away, like she says, you know, she returns our calls on time, even on the days when I'm feeling really bad .. she'll tell me what I need to do'.

[Female, Bangladeshi/British Bangladeshi, Muslim]

In a few cases, people (usually those who have been very unwell recently) spoke of how they could get an outpatient appointment immediately, or within a few days of requesting it.

Participants also spoke of how easy it was to talk members of the gastroenterology team, (including both nurses and doctors), often alluding to the personal nature of the conversations:

'I can tell her [IBD Specialist nurse] anything, and she is not embarrassed because she knows all about colitis, and so I am not embarrassed to tell her about the toilet and my problem'.

[Male, Indian/ British Indian, Hindu]

#### And

'He's really kind of, he's a good, he's a good doctor in that way, and he does loads of clinical trials and this and that, so he's always letting you know about stuff. Easy to approach, easy to talk to. .. Yeah, he's good. I can't, I can't say anything bad about him'.

[Female, British Indian, Hindu]

In terms of information giving, nurses and doctors appeared to have different roles, with participants referring to how it was easier to talk to nurses since they had more time available to explain things, and that they were good at giving practical advice, but that it was reassuring to speak to doctors about treatment because they had more expertise on this subject. Overall participants spoke positively about the information they were given about the condition and their treatment options, including information provided in written form, by DVD and verbally, often referring to the efforts some staff made to ensure that they fully understood the information provided:

'The doctors they were very good, you know, these consultants up at [name of hospital], they asked me if I had any questions and if I didn't really understood or if I'm not sure, if I'm worried, they talk it through, and if they started treatment they told me, you know, we're going to start so-and-so treatment, and this is how it works and what happens'.

[Female, Pakistani/British Pakistani, Muslim]

Participants explained that one of the main benefits for them in developing a better understanding of their condition and how to manage it, was that it gave them an increased sense of control over the disease. In some cases, discussions between staff and patients moved beyond information giving to providing emotional support, with a few participants explaining that staff played an important role in boosting their morale or relieving anxiety:

'She [nurse] told me to just stay calm and she just told me about the A to Z of colitis and its occurrence, how it affects your daily life, and I just .. it, it was a bit of a shock to my system to me, at that time. He [consultant] told me to just keep going. He, I think he came to see me every other day on the ward. He just told me to be brave and strong and be courageous and his support was invaluable really'.

[Male, Pakistani/British Pakistani, Muslim]

In a few cases people spoke of the gastroenterology team taking on an advocacy role, writing to employers who patients were having difficulties with to explain their health condition and its implications for work, and to other health specialists in an attempt to secure speedy access to IVF treatment. These efforts were greatly appreciated by patients, even though they had not been effective in getting people the help needed.

The value of peer support which had been organised or facilitated by the IBD specialist nurse was described by a number of participants. This included an online discussion group set up for patients by nurses, a nurse putting someone in touch with another patient the same age, and meeting patients through attending infliximab infusion clinics. The person who had meet others through an infliximab infusion clinic was particularly positive about the experience, explaining that it had become an important source of information and support for him, particularly as the IBD specialist nurse was at hand to provide advice. For further details on how this operates, and why it is a so beneficial, see Figure 1.

Figure 1: Example of infliximab infusion clinic acting as an informal support group for patients

'For me having these infusions done allows me to talk to other people in a similar situation. Okay, their Crohn's might be in a different area, but you've always got someone there who's ready to talk about it. Because you've got someone there who's medical [IBD Specialist Nurse], it's great because, you know, someone might say something like "Have you tried this?" And I'm like, "Oh yeah, that's a good idea, I'll try that." But my nurse might say, "Alright, be careful of those because your Crohn's is in a different place to that gentleman's Crohn's." She said, "You've got it in your colon, he's got it in his smaller intestine." So for example, we were talking peanuts and almonds and things like this. And I said, "I might try that" and she goes, "Oh, you could actually maybe try it," she said, "but because of where your Crohn's is crush them [nuts] down and try them," and, you know, you're not going to get that off the Crohn's and Colitis website. It's just not going to happen'.

[Male, Indian/British Indian, Sikh]

Although many people spoke of the excellent support provided by the IBD specialist nurse, it is worth noting that a few said it had taken them some time to realise what support was available, suggesting that in some cases more could be done to make this clear to patients:

'The IBD nurse gave me loads of information, access to .. NACC and .. information on lifestyle, drugs .. treatment .. issues with work and all that kind of stuff. I email her and say, you know, do you know what else I could take and, you know, and she'd get back to me and say, well try this, and so .. we have that sort of working dialogue. .. But not, it's not always been that way it's just been recently, and I think, I don't know whether that was me not understanding that she was there all the time or just feeling .. that that wasn't part of her role. It took me a while'.

[Female, Indian/British Indian, Hindu]

## Difficulties with the support provided by gastroenterology services

Although most participants were happy overall with the support provided by their gastroenterology team, there were areas of difficulty. Many of the difficulties people reported were about communication failures. Other concerns were about the focus of consultations being purely medical, the infrequency of outpatient appointments, medical errors and the shock of transfer from child to adult services.

## Communication failures

Problems with communication were wide ranging in nature. Some participants spoke of how they had been given no clear explanation as why they had IBD and of how difficult this was to come to terms with:

'Why is it happening? They don't have an answer and they can't give me an answer. They know how to treat it, they can use different things.... "Oh try this, oh no, this is better cos it's a different chemical. Oh okay, yeah, this works really well." So this is what basically these experts are doing. Yes, they're doing their best, I'm not saying they're not doing their best, you know, but it's something, you know, not in their hands, I would say the medical experts have their limitations in the sense they have not come to the conclusion or anything in terms of how they can treat this hundred percent. Yes, they have all these medications, and Infliximab, they have nothing else after that. But my question is "Why does this happen?" which they don't have an answer to. If they don't have answer they can't satisfy me, so I still have the question in my mind'.

[Male, Pakistani/British Pakistani, Muslim]

Sometimes it was that people felt they had not been given a full explanation of treatment decisions and it was notable that when people had not been given a clear

explanation, and they were unhappy with treatment decisions, this led to feelings of anger and frustration.

Just a few people complained about the way in which staff spoke to them, referring to them not being sympathetic or not listening to their health problems. Failure to feed back test results, usually to GPs, was mentioned by a small number of participants. In just once case, a person complained that the clinic had failed to inform her about test results. However, this was very distressing for the patient in question:

'I have given blood twice now and I haven't heard anything about the results. No letters or phone calls. A doctor should take care of their patients. Like I'm ill here, I keep thinking should I continue with these medications or not? The doctors should tell me. When a patient gives blood the doctor should tell them whether they should continue with the medication or not. I've heard nothing and it's been a month and a half now... Sometimes I tell my husband, if I could get someone to help me put a complaint in, I would do it. It just feels like we are crying because we are ill and they are trying to follow procedures and protocols'.

[Female, Pakistani/British Pakistani, Muslim]

## The focus of consultations is purely on medical treatment

The short time allocated to outpatient appointments meant people felt there was limited time to talk about anything other than medical treatment. A number of participants spoke of the focus of consultation being solely on medical treatment, with gastroenterology staff not taking the time to ask how they were coping with everyday life, or to understand how their health was impacting on them emotionally:

'They ask the question of what symptoms am I getting with my Crohn's where I can just waffle on and just say "Oh I've got diarrhoea" etc., etc. But it's not, the question's not there as to saying like "what are you finding difficulty with, how are you living at home or have you got someone supporting you at home?"

[Female, Bangladeshi/British Bangladeshi, Muslim]

One woman who had been dealing with depression due to ill health for many years, said she did not talk to her gastroenterology team about these issues, explaining:

'I mean, as brilliant as my Gastroenterologist is, they're there to prescribe the medicines, try and prescribe me something that will help, that will work'.

[Female, Indian/British Indian, no religion]

Another woman who was prescribed diazepam by her GP for panic attacks brought on by anxiety about travelling due to IBD symptoms, said her gastroenterology team were unaware she had these difficulties or was receiving this treatment. Others who had less severe mental health issues felt this was any area where staff could do more to help:

'I think both with Crohn's & Colitis UK and with the clinic there's not much emphasis on like mental wellbeing with like IBD .. cos I've found like during periods of flare-up I've been really, really depressed .. but I've not really done anything about it in terms of approaching my doctor or .. not that I want to be on antidepressants but, you know, just anything that they might be able to suggest or they might be able to say, you know, this might help or that might help. .. I've not, like they've never asked me about it and therefore I've never mentioned it to them. .. Yeah, I think maybe more emphasis needs to be put on that, because personally I think the psychological effects are much worse than the physical ones'.

[Indian/British Indian, Hindu]

There were also a number of complaints about lack of dietary advice. One man spoke of how, despite experiencing 'morning sickness' for many years, he had only been referred to a dietician once in the past seven years and was left to work out what he should and should not eat for himself. Another spoke of asking his consultant twice for a referral to a dietician, but he had been refused without explanation. In a rare example of a person being referred to a nutritionist, the advice offered was considered unsatisfactory as the dietician told the patient she was doing everything she should, which left the participant wondering whether the dietician knew much about the relationship between diet and IBD. A number of people spoke of health professionals' uncertainty about the links between diet and IBD, and this was clearly a source of frustration for some:

'IBD and diet, I feel, this is one of the things I do feel quite, quite strongly about, because every time you go .. and speak to a professional they, they'll tell you that there's no clear links between IBD and diet, but there has to be because, you know, you can't have, I mean I have met so many people with IBD who have issues with food intolerances, and even if it, for me I just want to know is it because the IBD triggers some changes that, that causes you to be intolerant or is it that food intolerance is part of IBD .. and no-one seems to be able to give us an answer'.

[Female, Indian/British Indian, Hindu]

As discussed in Chapter 5, some patients were using, or were interested in using some form of alternative or complementary medicine or therapy. In a number of instances, the gastroenterology team were aware of the treatments being used and had advised that it was safe to do so, but that it might not help a great deal. One woman said she had attempted to discuss such options with her team but had received such a negative response that she no longer initiated such discussions.

Others had not discussed the treatment with gastroenterology staff, feeling they were not interested in discussing anything but 'conventional treatment'.

## The infrequency of outpatient appointments

A number of people made complaints about the scheduling of appointments. A few said that infrequent appointments meant that, even when they had something important to discuss with the team, by the time their appointment came around they had often forgot about it. While some were aware they could contact a nurse in between appointments, others spoke of wanting to speak directly to the doctor:

'You don't always want to see the nurse specialist, you want to see the doctor, you want to see the doctor who's actually dealing with your case rather than just being passed down messages through your nurses'.

[Female, Bangladeshi/British Bangladeshi, Muslim]

Difficulty getting an emergency appointment was particularly upsetting when it was requested because of a flare up:

'I feel like I have no peace, when I am ill. In these circumstances you want them to see you really quickly and give you treatment to make you feel better. If you have to wait two weeks, you feel worse. We shouldn't have to wait 2 weeks. Not even a week. Doctors should feel caring towards patients and get appointments for them in one or two days. They have your medical records and they should see you quickly'.

[Female, Pakistani/British Pakistani, Muslim]

## Complaints about medical treatment

Complaints about medical treatment were rare but sometimes serious. They included: medication errors, usually by junior members of staff (in one case resulting in miscarriage); poor care both during and after a colonoscopy (attempts to persuade the patient not to have anaesthesia and lack of time to recover before being asked to leave the clinic); and being refused requested medication without a full explanation.

#### The shock of transfer from child to adult services

Given the background of the patients who took part in this study very few were in a position to speak about the transfer from children's to adult services. The one person who did speak about the issue, spoke of what a shock it had been to be an environment that was more nurse-led and where she was expected to make her own decisions about treatment options, leaving her feeling that her care was not being taken seriously.

## Views on how well services attend to the needs of South Asian patients

Study participants were asked if gastroenterology staff were aware of and responded to their needs as a South Asian woman/man. Many could not see how their needs were any different from the wider IBD patient population. Some seemed to interpret the question as being about whether they had experienced any form of discrimination, and most of these people reported that they had not:

'Many, many people waiting, some are Indian, people from different countries, also like other people, like white people, and the clinic is treating everyone just the same; if you come late, they say stay there, we're going to see you later. It doesn't matter your colour, or anything like that, they will help you, so it doesn't make any difference'.

[Male, India/British Indian, Hindu]

Only one participant felt there might be some small element of racial discrimination from doctors but said he did not want to say much more about this because he felt it might be held against him.

A number of South Asian patients not born in the UK spoke of their gratitude for the service they received, which they were aware was not always available in other countries. For example:

'I think we're very fortunate and lucky here in this country, that we have consultants and treatments available to us and are looked after quite well. Back home they can't afford to go to a hospital, because you have to pay a fee, even for your medication and stuff like that and then they, I'm sure, suffer big time and for that I feel really, really sorry for them .. and I'm really grateful that I'm here'.

[Male, Indian/British Indian, Sikh]

There were also examples of culturally sensitive practice, with a few participants speaking of being pleased with the guidance and support they received from their gastroenterologist prior to going abroad. One explained that she had been in constant contact with her consultant in the lead up to her trip as she was taking immunosuppressants at the time and needed help with arranging to have the appropriate vaccinations. Another spoke of her consultant helping her overcome her fear of going on holiday by giving her additional medication in case of a flare up, clear instructions on how to take the medication and an email address so that she could get in touch with the clinic should she run into any difficulties whilst abroad. One person spoke positively about consultant warning her that, as she is South Asian, she is more likely than other patients to be deficient in Vitamin D.

A few people did highlight shortcomings in service provision to South Asian patients and these feel into two main areas: (1) not appreciating the importance of being able to fulfil cultural expectations of what constitutes a good wife, mother or husband; (2) not providing adequate translation services to patients whose first language is not English.

With regard to not appreciating the importance of being able to fulfil cultural expectations of what constitutes a good wife, mother or husband, as discussed in Chapter 5, some participants felt that having IBD impinged on their ability to fulfil these roles. This was a topic that two people returned to when discussing support from gastroenterology staff, feeling that staff did not fully appreciate the importance of providing them with support that enabled them to fulfil these roles:

'For an Asian woman there are a lot of responsibilities. Taking care of the children and her husband and her home. According to my responsibilities I don't feel like I am receiving the best treatment. What should I focus on, my children, husband or the disease?'

[Female, Pakistani/British Pakistani, Muslim]

#### And

'I'm just now with the issue with the pregnancy and stuff, cos obviously I've been pushing it towards them saying, like it's important that I need to get pregnant and they don't understand. ... it's a big stress thing for me in the fact that I'm not getting pregnant and I want to get pregnant.'

[Female, Bangladeshi/ British Bangladeshi, Muslim]

With regard to the provision of adequate translation services, we have limited evidence from this study on the experiences of South Asian patients who do not speak English since only two participants were in this position. However, interview data from one of these participants clearly illustrated how difficult it can be to access support when you do not speak English, particularly when the clinic fails to provide an interpreter:

'I can't speak English with the white doctors. Sometimes an interpreter is at the appointment but for the past 3 or 4 appointments no one was booked...... Sometimes the hospital tells me to wait for the interpreter. I have waited a couple of times but if I wait then where do I leave the kids? My husband normally goes with me. We need appointments at times that suit us. My husband works from 7am to 3pm. Then it's time to pick up the kids. If I wait for the interpreter and find someone to take care of the kids, they will not look after the younger children for the whole day. The hospital should be aware of these issues and make allowances in how they allocate appointments. Yes, I am ill and so I should wait, but I can't do this all the time'.

[Female, Pakistani/British Pakistani, Muslim]

She goes on to explain that she relies on her 12 year old daughter to look up information about IBD and the treatment she is receiving on the internet, and then translate it into Mirpuri for her. Her daughter tries to reassure her that she will be okay.

A participant who spoke English herself, but who had a cousin with IBD who does not, spoke of how she has recently begun accompanying this cousin to clinic appointments because he did not have an adequate interpreter. She thought her involvement had helped her cousin to understand the importance of adhering to medication.

Other participants who spoke fluent English said they had noticed South Asian patients attending clinic who the nurses did not appear to be able to communicate with. Some commented that although there is written information on IBD available within clinics, they had never noticed any in South Asian languages. Lack of translated materials was considered a particular problem for people who speak Punjabi as there are no words for IBD in this language and knowledge of the condition was considered to be particularly limited in this community. One person warned that, given that she speaks English and it had still taken her some time to understand the support provided by the IBD specialist nurse, patients who do not speak English are likely to be particularly vulnerable to missing out on the support that is available.

## Additional support wanted from health services

Participants made a number of suggestions regarding the additional support they would like to see provided by health services, most of which fit with the difficulties with services outlined above. These are listed below.

## **Gastroenterology services:**

- Advice on alternative therapies.
- Access to a dietician.
- Access to a counsellor who specialises in supporting people with medical conditions.
- Help making a successful application for disability benefits.
- A private room when first diagnosed so there is easy access to a toilet and privacy.
- To be given a choice over which member of the team you see.
- Clinics for teenagers.

#### GPs:

- GPs need to have a better understanding of IBD in order to: (1) prevent delays in diagnosis; (2) ensure a more empathetic GP service.
- Access to faecal calprotectin and blood tests via GP surgeries rather than having to travel to hospital.

## **South Asian specific recommendations**

While the small number of shortcomings in service provision for South Asian patients point to ways in which services could be improved for people with this background, very few participants made clear recommendations as to how to develop services. One participant felt that setting up a peer support group specifically for South Asian patients might be useful as people would be able to talk to someone who understands their lifestyle. Another person felt that South Asian patients need dietary advice which takes into account the kind of food that people from a South Asian background eat.

# Chapter 7 Views on Crohn's and Colitis UK and the Support it Provides

# Introduction

This chapter begins by reporting on participants' awareness of, and engagement with, Crohn's and Colitis UK, before describing views on the support offered by the organisation. We then go on to outline perceived cultural barriers to South Asian patients' engaging with the organisation, and recommendations as to how Crohn's and Colitis UK can better support South Asian patients.

# Awareness of Crohn's and Colitis UK

Knowledge and understanding of the support offered by Crohn's and Colitis UK was, typically, very low. Only a small minority identified themselves as members and even fewer actively used the organisation in an on-going way. Indeed, for many, the research interview (particularly the use of the visual aid depicting Crohn's and Colitis UK's services and support, see Appendix 2) proved to be educational in terms of demonstrating to participants the different ways Crohn's and Colitis UK seeks to support patients and families.

However, it is important to stress at the outset that there were some very positive accounts of the support Crohn's and Colitis UK provides from some of those who had actively sought information and advice:

"...they're very good. They give you information, they sent me the IBD book....and the lady I spoke to...they are helpful".

[Female, Indian/British Indian, Sikh]

# And

'I think the website is full of good information. And, you know, if you're going through a bad, bad stage, you can speak confidentially to someone who will give support and help'.

[Male, Pakistani/British Pakistani, Muslim]

One interviewee particularly applauded Crohn's and Colitis UK's presence on social media and the way this generated 'open' conversations about IBD:

'They're starting to get on social media now which I think it's really good...like they're on Facebook, Twitter and everything. And I think on Twitter especially I've found loads of people just talk about IBD really openly'.

[Female, Indian/British Indian, Hindu]

Overall, the form of support most consistently identified as Crohn's and Colitis UK's core function was as an information source. Many recalled Crohn's and Colitis UK leaflets being given to them around the time of diagnosis or seeing them in clinic subsequently. It is important to note here that despite the popularity of social media, not everyone wanted on-line resources. For example, some had stopped reading the Crohn's and Colitis UK newsletter when it was delivered as an email attachment rather than through the post:

'...I'm not very good with email, like computer, I prefer paper'.

[Female, Indian/British Indian, Hindu]

# Reasons why individuals did not engage with Crohn's and Colitis UK

A key reason for not engaging with Crohn's and Colitis UK which emerged from the interview data was the episodic nature of the condition. During periods of good health interviewees spoke about not perceiving there to be a need to draw on the support Crohn's and Colitis UK might have to offer them. At the same time, the often debilitating symptoms during a flare up meant interviewees did not have the energy or motivation to explore Crohn's and Colitis UK as a source of a support or use its services.

In addition, even after being shown the Visual Aid depicting Crohn's and Colitis UK's services and support, some interviewees did not perceive Crohn's and Colitis UK as being able to offer any new or additional information to that which they received from clinic staff and/or family and friends who had a medical background and/or experiences of IBD themselves. It is important to remember here that interviewees were consistently very positive in their accounts of the specialist IBD team caring for them. It is worth noting, however, that Crohn's and Colitis UK was viewed as having a potential role in terms of providing emotional support – something which clinics were typically viewed as not having the resources to offer. Indeed a minority had used the Crohn's and Colitis UK helpline in this way.

A further reason for not using the Crohn's and Colitis UK as an information source was that it did not address, in the way that was wanted, one of the dominant information needs identified by our interviewees; namely, a range of 'uncensored' suggestions and advice on new dietary regimes and CAMs that they might try. A number of participants said they favoured using unofficial forums instead:

'On the non-official forum you get the voice of real people. People who are actually suffering with it, and their view might not be the same as NICE. On Crohn's and Colitis UK website they have to be careful what they say and that's

not useful. ...Unofficial websites are more honest – people tell you the truth'. [Male, Indian/British Indian, Sikh]

A handful of interviewees had attended, or were a regular member of, a support group. Typically they described this as a positive experience:

'I went to the meeting. I found it really good just to talk to people. I didn't really talk to many people, but just listening to people and hearing the medical advisor. Listening and thinking, "Oh, I'm not alone". There's other people who have got this'.

[Male, Indian/British Indian, Hindu]

#### And

'I know I am with like-minded people. That I don't have to explain everything to. And it has helped a bit knowing that everyone has had the same kind of issues as I have had'.

[Female, Indian/British Indian, Hindu]

Among those who had *not* linked up with a local Crohn's and Colitis UK, there were mixed feelings about the benefits of attending something like this. Some expressed a disinterest in other people's experiences and stories and saw it having no benefit to themselves. Indeed, some spoke of wanting to avoid speaking with others with the same condition in case they heard about experiences that were worse than their own. Interestingly, those participants who had had the opportunity to talk to other patients with IBD in a clinic setting, or through pre-existing friendships with people who had happened to have IBD, had found this a helpful experience. Furthermore, a number of participants who opted to be interviewed by a LISA project interviewer who had personal experience of IBD spoke of how enjoyable an experience this had been, with many taking the opportunity before or after the interview to ask the interviewer how they managed their IBD.

# Do cultural issues impact on South Asian patients' engagement with Crohn's and Colitis UK?

We examined the interview data to see if participants located their reasons for a lack of use, or engagement with Crohn's and Colitis UK on cultural or religious factors. Participants offered views on this with respect to their own situation and in terms of their ethnic group more generally. At the outset, however, it is important to note that some interviewees seemed to expect that Crohn's and Colitis UK would know they had IBD and contact them directly to offer support perhaps, in part, flagging up a lack of understanding that voluntary sector organisations operate separately to the NHS:

'I thought they would have made themselves known... written to me...when I became a Crohn's disease patient'.

[Female, Indian/British Indian, Hindu]

#### Barriers to use of on-line resources

Overall, access and use of the internet and social media varied enormously within our sample. Many had not viewed Crohn's and Colitis UK's website and, as noted above, there are a range of reasons why individuals may not regard Crohn's and Colitis UK as a necessary or desired source of support.

In addition, however, some of the older people interviewed reported they were unfamiliar with using the internet and social media and would not use these media to access information and support regarding their IBD. Other interviewees also noted that the 'older generation' were not 'on-line' and that this, combined sometimes with literacy difficulties, were barriers to older people using Crohn's and Colitis UK's on-line resources.

# The representation of South Asian individuals in publicity and information resources

Whilst not a personal barrier to using Crohn's and Colitis UK as a source of support, the observation was made that Crohn's and Colitis UK should review the presentation of minority groups per se in its printed materials and its website:

'I really don't think there's anything like specifically for South Asians that I saw on the website or in newsletters or anything. There really does obviously need to be more work done with South Asians'.

[Female, Pakistani/British Pakistani, Muslim]

# **Support groups**

A small minority of participants had attended or were regular members of a local Crohn's and Colitis UK support group. We noted earlier the reluctance of some to hear other people's stories and an ambivalence to engage with support related to IBD during periods of good health. These in themselves were, for some, barriers to joining a support group. In addition, those who struggled with on-going 'low level' symptoms such as tiredness, or issues such as low mood, identified these as barriers to joining a support group. Others said family/caring responsibilities meant they were unable to get along to a group. However, among the individuals we interviewed, none of these barriers were explicitly linked to South Asian identity.

Over and above these, however, certain (perceived) features of support groups were identified as barriers to patients of South Asian origin attending a support group. As with other issues, sometimes interviewees, whilst not viewing something as a barrier themselves, believed they could be for some individuals with the same racial/ethnic background. First, support groups could be perceived as for 'white, middle class' people. It was noted that, for some, going to a meeting where there were no other South Asian people in attendance would be an alien and uncomfortable experience:

"...in this country a lot of Asian people tend to socialise with other Asian people and they talk to other Asian people, go to Asian shops and everything...so it might seem strange to them to go somewhere where there's no Asians at all. They need to know they aren't gonna be the only brown face in the room'.

[Female, Indian/British Indian, Hindu]

Second, using pubs as venues for meetings was a barrier for some. A familiar venue, such as the hospital where the clinic was held, was preferred. Meeting in a pub also, implicitly, suggested a lack of racial/cultural awareness among the organisers of a group. It was noted that greater representation of South Asians among those organising local groups, and where South Asians were involved, making this explicit in publicity, were two strategies which could address these issues.

A few participants expressed the view that South Asian people are more 'private' than other ethnic groups. The notion of sharing experiences in a group, particularly perhaps regarding a culturally sensitive aspect of body function, and with people outside of their own community, was an alien concept. It is essential to note here that some individuals who had held such views in the past, through conversations with other patients at 'infusion clinics', now saw the value of peer support.

# Support in languages other than English

Not unexpectedly, the importance of making Crohn's and Colitis UK publications available in South Asian languages was noted. However, a couple of participants noted that literacy in any language would be an issue for a minority of South Asian patients. Here older patients were identified as being most likely to have issues around language and literacy. Importantly, a number of interviewees expressed the view that patients who did not speak English, or had English as a second language, were particularly vulnerable to unmet support needs as their ability to draw on support and advice during clinic appointments was likely to be limited to some degree. The suggestion was made the information resources should be specifically written for this group.

In addition, one participant predicted that the Crohn's and Colitis UK helpline would be unlikely to be able to assist if someone whose first language was not English got in touch:

'Being realistic, would Crohn's and Colitis UK really be able to help someone who can't speak English with applying for benefits? Seems unlikely. Imagine they would be put off at first phone call to the organisation when it becomes apparent that they don't have people to respond to enquirers who speak South Asian languages'.

[Female, Pakistani/British Pakistani, Muslim]

# Views on how Crohn's and Colitis UK can better support South Asian patients

In addition to addressing the barriers to using Crohn's and Colitis UK outlined above, three further areas of work or activity were identified by interviewees:

- Awareness raising about IBD in South Asian communities.
- Supporting the wider family.
- Addressing specific information needs.

# Awareness raising about IBD in South Asian communities

When asked about how Crohn's and Colitis UK could improve its support to South Asian patients, one of the dominant themes to emerge was the role of awareness-raising about IBD within South Asian communities. Interviewees sometimes gave the example of the awareness-raising activities regarding diabetes, high cholesterol, and alcohol, which they believed had led to people talking more openly about these health problems and needed to be replicated for IBD. They called for publicity and information about IBD to be widely distributed in a range of different settings such as GP surgeries, places where South Asian people go to socialise, and temples and mosques. One participant suggested using South Asian 'personalities/celebrities' to promote this work. Another mooted the idea of workshops or information events as well as printed information and publicity. A number noted the need for this information to be provided in a range of languages.

# Supporting the information needs of the wider family

Another sphere of information and awareness-raising, requiring more specific and detailed information, was to close and wider family members of patients, and including family members living overseas. The need for materials to be tailored to the particular audience, which took account of cultural and/or religious influences on beliefs about health, illness and diet, was noted.

# Addressing specific information needs

Lastly, the following areas of information need – all perceived to be specific to South Asian patients <u>or</u> requiring the provision of information which was culturally appropriate – were identified:

- travelling to South Asian countries, including:
  - managing long journeys;
  - dealing with flare-ups, including use of local health care;
  - o transporting and storing medication.
- South Asian food/diet and IBD.
- complementary and alternative medicines (CAMs) commonly used within South Asian communities, including advice on effectiveness and warnings if likely to cause complications, or be unsafe when combined with IBD medications.

It was clear that as well as being useful resources which Crohn's and Colitis UK could produce (in written and audio-visual formats), these were also information needs which clinics should be aware of.

Finally, and on the face of it counter to the resistance to hearing other patients views and experiences noted earlier, a number of interviewees noted they would value hearing about how others coped with living with IBD on a day to day basis. Audiovisual resources or the opportunity to speak with someone else with the condition (but not requiring attendance of a support group) were both suggested as effective ways of addressing these sorts of information needs.

# **Chapter 8** Discussion

# Introduction

In this final chapter of the report, we summarise the research findings and reflect on what they add to the existing literature on the experiences and support needs of adults with IBD, before outlining the implications for both Crohn's and Colitis UK and gastroenterology services. Finally, we discuss the strengths and limitations of the study.

# Summary of the findings and what they add to the literature

This is the first study to give South Asian adults in England an opportunity to talk about their experience of living with IBD. The research confirms there is a generic experience of IBD, irrespective of ethnic background. As participants explained, much of their experience was driven by IBD symptoms, and these symptoms are common to IBD patients of all backgrounds. IBD flare ups were painful, exhausting and involved frequent trips to the toilet, so many people remained housebound when experiencing them. In addition to this, in between flare ups, many people experienced ongoing milder symptoms, as well as anxieties that such symptoms might indicate the onset of a flare up.

As in the wider IBD literature (McCormick et al., 2012; Hall, 2005), this study found that people were making considerable efforts to control or cope with physical symptoms of IBD. The majority of participants saw medication as necessary for preventing flare ups and controlling symptoms, despite having concerns about side effects, but they also used a wide range of other strategies, some of which were driven by beliefs that IBD is exacerbated by diet and stress. Practical and emotional support was provided by immediate family. While members of the extended family were often aware of the person's health difficulties, they were often unable to help due to living too far away or having their own caring, or other, responsibilities. Despite the efforts people were making, and the support received, having IBD often took its toll on people emotionally. Feeling low, depressed or anxious added further to the difficulties people experienced so that, even when symptom free, some people curtailed their activities due to loss of interest or worry about the possibility of experiencing symptoms. As a result of this combination of physical symptoms and emotional difficulties, people reported struggling at times with maintaining a social life and friendships, completing higher education/training courses, managing work, caring for children and in relationships with partners. A few people also reported financial difficulties as a consequence of giving up work/reducing their working

hours and/or their partner having to stop work/reduce working hours in order to care for children.

In relation to health services, once diagnosed, very few consulted their GP for advice or treatment. Most often this was because they felt their GP did not have sufficient knowledge about IBD and/or they did not have up to date information about the person's health status. Sometimes trust in a GP had been broken by the GP being slow to make referrals to gastroenterology services when the person first became unwell, resulting in delays to diagnosis, an issue familiar within the IBD literature and which efforts are being made to address (IBD Standards 2013 Update)

On the other hand, experiences of gastroenterology services, and relationships with staff, were overwhelmingly positive, with just a few exceptions. Participants appreciated the ease of access to the team outside of scheduled appointments, and some spoke of how comfortable they felt talking to staff about sensitive issues, such as bowel related symptoms. The information provided by IBD services was generally considered good, giving people an increased sense of control over their condition. In a few cases, the team had taken on an advocacy role and/or facilitated peer support, which was much appreciated. Nevertheless there were areas for improvement. These were largely around the feeling that the focus of consultations was purely on medical treatment, with staff not having or taking the time to discuss how they were coping with the condition; failing to provide dietary advice, or respond to requests to see a dietician. Many patients were using or interested in using Complementary and Alternative Medicines (CAMs) but a number had not discussed these options with the team, feeling that they were not interested or were 'negative' about anything aside from conventional medical treatments. This is of concern given that a number had suffered side effects from CAMs and it was only at this point that they learnt that some CAMS were best avoided when taking IBD medication.

All these findings fit with concerns expressed recently by a number of clinicians and academics writing about IBD services. For example, Andrews et al. (2010) has suggested that the medical management is increasingly being directed at acute flare ups of the disease at the expense of other aspects of care. Further, there is a danger with the development of new biological therapies that management will be become even more narrowly focussed, failing to address the psychological burden of living with IBD, quality of life, and specific psychological comorbidities, along with others issues. Others have also called for psychological care to be incorporated into the management of IBD (Mikocka-Walus et al., 2012; Moser, 2006). Unfortunately, within the UK providing such support may be a challenge. The fourth round of the UK IBD Audit (Royal College of Physicians, 2014) found that around a third of services routinely gave patients information on how to access counselling from a primary care or a third sector organisation, and only one in ten had access to a psychologist via a defined referral pathway. This report concluded that psychological support needs to be given greater priority at a local level and this could be helped by

the sharing of good practice by the small number sites that have managed to establish a dedicated service.

Knowledge and understanding of Crohn's and Colitis UK was typically very low. Key reasons why people did not engage with the organisation which could apply to people with IBD from any ethnic background were: the episodic nature of the condition, with people reporting not being motivated to use the organisation when well and being too ill to do so when having a flare up; they believed they already had all the information and support they needed, largely through gastroenterology clinics, but also sometimes through family members who were health professionals and friends who had IBD; and the organisation did not provide the 'uncensored' information about dietary regimes and CAMs which, although untested, they might like to try.

Although there was a generic experience of IBD, the research also revealed that aspects of South Asian identity and relationships with others in the South Asian community, did affect experiences. A number of factors were identified which influenced both how family, friends and the wider community responded to the person having IBD, and how participants themselves felt about having the condition. Participants reported that there were a number of culturally specific difficulties understanding IBD (e.g. people have never heard of the condition, there is no word for Crohn's in some South Asian languages, the word 'disease' has different connotations in South Asian communities etc.). Some said they withheld details of their diagnosis from all but their immediate family. This was primarily because ill health was particularly stigmatised in South Asian communities due to the perception that it impacted negatively on 'marriageability' and caused concerns for the health of future offspring. Furthermore, while it was acknowledged that talking about IBD symptoms (particularly bowel-related) might be embarrassing for many people, there was a perception that this subject was particularly taboo within South Asian communities, further adding to the culture of silence around IBD. While researchers investigating the experiences of the wider IBD have found that people tend to see their illness as embarrassing, taboo and often misunderstood by others populations (Hall, 2005), taken together these findings from the LISA Project suggest that such feelings may be amplified in South Asian patients.

In their study of younger South Asian adults, Nash et al.'s (2011) identified that being part of a culture in which eating spicy food is the norm can cause difficulties for people with IBD. Similarly in the LISA Project most participants believed that such food exacerbated IBD symptoms and avoided it, which could mean eating different food from other members of the household. This was manageable when the person was in charge of family meals and cooking, but more difficult if they were not (i.e. young person living with parents, a woman living with in-laws). A reluctance to eat/avoidance of spicy food was also an issue when out socially, particularly when participants had chosen not to tell others they had IBD. Some dealt with this by

avoiding social events. Occasionally the difficulty of refusing food, or social pressure to eat, either in or outside the home, meant that people ate despite knowing it would make them unwell. An additional food related issue, not identified in Nash's study, were beliefs within the South Asian community about the links between food and health. In some religions, some foods are blessed and there are beliefs that ill health is brought about by an imbalance in foods. Such beliefs made it difficult for some individuals to get others to accept their preferred diet.

Moody et al. (1998) has suggested that South Asian patients may live in areas of the country where CAMs, particularly South Asian CAMs, are widely available and may therefore be more likely to use them. As this was a qualitative study we did not attempt to examine the extent of CAMs use amongst participants. However, the research did find that people had mixed views on CAMs. While some had used them and found them beneficial, others were more sceptical, particularly when they had family members who were health professionals and had warned them not to use 'untested' treatments.

Participants also spoke of expectations within their community as to the role of women and men and its impact on their experiences of living with IBD. For women, this mostly concerned being a mother and homemaker. For men, it was about financial responsibilities. While men and woman from any background might worry about their ability to undertake gender orientated roles when ill, in this study some of the women were also concerned about the social consequences of not being able to do so, feeling that they would not be forgiven by others in their community if they failed to take on the duties expected of a wife and mother. These are not issues found in other research investigating the experiences of white British IBD patients in their parenting role (Mukherjee et al., 2002).

Making visits to extended family overseas was something that a substantial proportion of participants did on a regular or occasional basis. While people often worried about making these trips due to concerns about travelling whilst experiencing bowel related symptoms and limited access to toilets, for many they were worthwhile as it allowed them to attend important family events. These trips often involved stays of a few months, during which time there was potential for fluctuations in people's condition. Many were pleased to find their health improved while away, which they accounted for as being a consequence of a warmer climate and being cared for by family members. Where people did experience flare ups, there were often clear reasons for this, such as not adhering to medication or changes to diet.

Many people reported deriving a great deal of benefit from their faith, describing ways in which it helped them to cope with having IBD. However, Muslim participants spoke of difficulties praying because it involved a physical act which they were sometimes not well enough to perform. Despite knowing that praying when unwell

was not a requirement of their faith, not being able to do so was a source of great upset to some. Furthermore, Muslim patients explained that when experiencing bowel related symptoms they often limited the amount of time spent away from home due to few public toilets having the amenities needed for them to clean themselves with water, as required by their faith.

Finally, a small number of participants felt living with extended family was difficult due to the need to share (limited) toilet facilities. Embarrassment about bowel related symptoms was also an issue when people first moved in with extended family.

In 2012, a UK gastroenterologist published a paper commenting on the provision of gastroenterology services to South Asian patients in the UK (specifically Panjabi patients), stating that the availability of appropriate services is woeful and yet seldom leads to formal complaints. He expressed particular concern about the lack of adequate translators, and failure to provide translated versions of information literature (Mayberry et al., 2012). Our research found that, although important, translation is just one of many issues that gastroenterology services need to be aware of when supporting South Asian patients. Typically when asked about their needs as a South Asian patient, people were simply appreciative of the care they received and could not identify ways in which their needs might be any different from other IBD patients. It is important to note that there were also examples of culturally sensitive practice. For example, consultants taking time to help prepare people for trips overseas, and advising patients about the increased risk of vitamin D deficiency. Where shortcomings did exist, not providing adequate translation services to patients whose first language is not English were mentioned, but concerns were also expressed about gastroenterology staff not appreciating the importance to (some) South Asian patients of being able to fulfil cultural expectations of what constitutes a good wife, mother or husband, and the need to find treatments and/or strategies which enabled them to take on these roles.

A number of South Asian-specific barriers to engagement with Crohn's and Colitis UK were also identified. It was suggested that the older generation of South Asian patients are unlikely to be 'on line' and/or necessarily able to read English, making it hard for them to access the Crohn's and Colitis UK website. Concerns were expressed some years ago that South Asian do not attend local Crohn's and Colitis UK meetings (Moody et al., 1993). Participants in the LISA project identified a number of barriers to attendance. These included: the perception that such meetings would mostly be attended by white middle class people and that being the only South Asian person present would be uncomfortable; the use of culturally inappropriate venues such (i.e. pubs); and that South Asian people are relatively private so would be uncomfortable sharing experiences in a group. Finally, the fact that most of Crohn's and Colitis UK's publications are only available in English and

telephone helplines are unlikely to be manned by people who speak South Asian languages were highlighted as shortcomings that might need to be addressed.

# Implications for practice

Participants who took part in this study were a diverse group and differed considerably in the extent to which they saw themselves as South Asian or engaged with the wider South Asian community. Many of the experiences and views outlined in this report were experienced by some, but not all participants. It follows that when reading this report it should not be assumed that the experiences reported here will apply to all South Asian patients, and/or across all South Asian ethnic groups. Instead, an individualised approach to care is needed, in which practitioners take the time to explore the particular social and cultural context in which individual patients live their lives and the support that fits with these circumstances. Nonetheless, taken together, the findings from this study do suggest ways in which organisations might be able to the improve support to South Asian people in general. Some of the issues identified are generic, that it is developing support would benefit both the South Asian and the wider IBD population. Others are specifically about the South Asian population. As will be seen, most of these issues need to be taken on by both Crohn's and Colitis UK and gastroenterology services, though their roles in developing or providing support may differ.

# **Psychological Support**

Given that disease remission alone is insufficient to improve patients' quality of life and alleviate their worries and concerns, many researchers have called for health professionals to take a more holistic or integrated approach to care, attending to the psychological burdens of the disease and patients' quality of life when developing treatment strategies (Mikocka-Walus et al., 2012; Moser, 2006; Hall et al., 2005; Casati et al., 2000). It is clear from the findings from this study that South Asian patients would also benefit from more psychological support. This needs to include:

- Access to psychological therapies for those experiencing feelings of depression and/or anxiety brought on or exacerbated by having IBD.
- Opportunities for people to discuss how they are managing their everyday life, and any strategies that might be helpful.

Given time and resource limitations, and that psychological support falls outside of gastroenterology staff's areas of expertise, it seems unlikely that such issues can or will be addressed within gastroenterology clinic appointments. However, it might be possible for teams to develop links with other services who can offer such support. Furthermore, patients may benefit from having the opportunity to talk to others who IBD about how they manage their condition. While difficulties getting people to attend 'support groups' are well known, the example given in this study of people

swapping experiences and ideas whilst attending an infliximab infusion clinic, with facilitation by an IBD specialist nurse, is one that other clinics might want to consider pursuing.

# Information on the usefulness and safety of CAMs for IBD

Many people were either using CAMs, interested in using CAMs or had been advised to use CAMs by family or friends. Given that some CAMs can cause complications for patients, particularly when combined with IBD medications, and that many seemed unaware of this, it is important that Crohn's and Colitis UK and/or gastroenterology services offer advice in this area. Without such advice, people will continue to look to unofficial online forums for new treatments and remedies and conceal this from their consultant and other clinic staff.

# Dietary advice which is culturally appropriate

The possible link between South Asian diet and IBD was clearly an issue for people, with many avoiding all spicy food in an attempt to control symptoms. Others spoke of fasting on a regular basis after fasting for religious reasons and finding that it helped. While limited availability to dieticians is likely to make provision of individualised dietary advice difficult, it should be possible to produce information leaflets on diet which are tailored to the needs of South Asian patients, covering foods typically included in their diet, but also advice on issues such as fasting which may be of particular interest to Muslim patients.

# Awareness raising about IBD within the South Asian communities and families

Taken together, the findings on how cultural factors influence both how South Asian people who have IBD, and those around them, respond to the condition suggest that awareness raising about IBD should be a priority. Any awareness raising needs to:

- Dispel myths that may exist about IBD (i.e. suggestions that it might be infectious, and/or life-threatening).
- Clarify the relationship between IBD and diet. In particular, it should be made clear that it cannot be 'cured' by eating particular foods, but that people may benefit from following a particular dietary regime, including the avoidance of spicy food.
- IBD symptoms and the possibility that during flare ups people may need rest and time out from usual roles and responsibilities.

Given the reported stigma around having an illness which children might inherit, any information provided about the causes of IBD, and a possible genetic component, needs to be dealt with sensitively.

Awareness raising needs to happen both at a community and family level. It would seem appropriate for Crohn's and Colitis UK to take on awareness raising at a community level, and indeed research participants recommended the approach taken on by other health charities who were reported to have visited temples and mosques to distribute information leaflets and give talks on the condition they represented.

In addition, there is a need for awareness raising at a family level, ensuring that those who live with the person with IBD, and other close family members, really do understand the condition, treatment regimes, and the help and support the person with IBD may need. Given that this requires an individualised approach, input from the gastroenterology team is likely to be needed. It is hard to prescribe how clinics should take this on as much will depend on the resources they have available to them. However, a good starting point might be welcoming family members to clinic appointments so that they have an opportunity to hear the information and advice being passed on to patients and ask questions. Crohn's and Colitis UK could support awareness raising at a family level by providing generic information resources which cover culturally relevant topics and are available in South Asian language, as well as occasionally running local meetings to which family members are invited.

# The need for advice and support on making trips to visit family overseas

The trips people made to see extended family overseas often involved stays of a few months, during which time there was potential for fluctuations in people's condition. These trips have implications for both Crohn's and Colitis UK and gastroenterology services since the data suggest that people need support with: being adequately prepared for making the journey, including advice on how best to store medication when travelling; advice on how best to deal with a flare up should this occur while overseas; reminders as to the importance of adhering to prescribed medication and usual diet; and, in case people needed medical attention while overseas, a brief written summary of the person's diagnosis and the treatment they are receiving the UK which can be given to medical staff.

# Increasing the visibility of South Asian members of Crohn's and Colitis UK

While people welcomed the LISA project, signifying an interest in South Asian people, there was a clear sense that Crohn's and Colitis UK is an organisation run by and for white British people. For many this was reflected in the lack of South Asian presence on publicity materials, publications or the website, and at local meetings.

This lack of visibility needs to be addressed so that South Asian individuals feel comfortable engaging with the organisation at either a local or national level.

# Strengths and limitations of the study

Given that the study involves patients known to be hard to involve in research, we are extremely pleased with recruitment to the study. Indeed, we exceeded our recruitment target and could have interviewed more individuals had time and resources allowed. The sample included 19 people who described themselves as Indian/British Indian, nine as Pakistani/British Pakistani, four as Bangladeshi/British Bangladeshi and one as British Punjabi. In terms of faith, there were 17 Muslims, nine Hindus and seven Sikh participants. Such a sample reflects both the ethnic background and the spread of faith groups within the UK population (Office for National Statistics, 2013; Office of National Statistics, 2012). Furthermore, both men and women were represented, as were all age groups, with the exception of people over 65 years of age. It did appear that giving patients the option of being interviewed by someone with personal experience of IBD was beneficial, with some participants reporting that it was one of the reasons they had decided to take part. It also resulted in some very frank discussions about experiences. As most of the participants in this study spoke English, we have limited information on the experiences of South Asian people for whom language barriers are an issue. However, discussions with gastroenterology staff in the clinics we recruited from indicated that the English language abilities of our sample reflected their experience of South Asian patients attending clinic, with them reporting only a very small proportion (5-10 per cent) required an interpreter present during appointments.

# References

El-Wakil A (2011) Observations of the popularity and religious significance of blood-cupping (al-ḥijāma) as an Islamic medicine, *Contemporary Islamic Studies*, 2. doi:10.5339/cis.2011.2.

Andrews JM, Mountifield RE, Van Langenberg DR, Bampton PA, Holtmann GJ (2010) Un-promoted issues in inflammatory bowel disease: opportunities to optimize care, *Internal Medicine Journal*, 40, 3, 173-182.

Atkin K (2009) Negotiating ethnic identities and health, in Graham H (ed.), *Understanding Health Inequalities* (2<sup>nd</sup> edition), Maidenhead: Open University Press.

Atkin K, Chattoo S (2006) Approaches to conducting qualitative research in ethnically diverse populations, in Nazroo J (ed.), *Methodological Issues in Research Relating to Black and Minority Ethnic Groups: Publishing the evidence*, London: Taylor Francis.

Beresford B, Greco V, Clarke S, Sutherland H (2008) An evaluation of specialist mental health services for deaf children, *Research Works*, 2008-2, York: Social Policy Research Unit, University of York.

Betancourt JR, Green AR, Carrillo JE, Ananeh-Firempong O (2003) Defining cultural competence: a practical framework for addressing racial/ethnic disparities in health and health care, *Public Health Reports*, 118, 293-302.

Borgaonkar MR, Irvine EJ (2000) Quality of life measurement in gastrointestinal and liver disorders, *Gut*, 47, 444-454.

Borgaonkar, MR, Townson G, Donnelly M, and Irvine EJ (2002) Providing disease-related information worsens health-related quality of life in Inflammatory Bowel Disease, *Inflamm Bowel Dis*, 8, 4, 264-269.

Cairns, SR, Scholefield JH, Steele RJ, Dunlop MG, Thomas HJW, Evans GD, Eaden JA *et al.* (2010) Guidelines for colorectal cancer screening and surveillance in moderate and high risk groups (Update from 2002), *Gut*, 59, 5, 1 May, 666-689.

Casati J, Toner BB, De Rooy EC, Drossman D, Maunder R (2000) Concerns of patients with inflammatory bowel disease: a review of emerging themes, *Digestive Diseases and Sciences*, 45, 26-31.

Conroy SP, Mayberry JF (2001) Patient information booklets for Asian patients with ulcerative colitis, *Public Health*, 115, 6, 418-420.

Craig G, Atkin K, Chattoo S, Flynn R (2012) *Understanding Race and Ethnicity: Theory, history, policy and practice,* Bristol: Policy Press.

Crohn's and Colitis UK (2014) *Inflammatory Bowel Disease and the Retirement Decision*, Report commissioned by Crohn's and Colitis UK.

Czuber-Dochan W *et al.* (2013) The experience of fatigue in people with inflammatory bowel disease: an exploratory study, *Journal of Advanced Nursing*, 69, 9, 1987-1999.

Drossman D (1988) Psychological aspects of ulcerative colitis and Crohn's disease. In *Inflammatory Bowel Disease*, Kirsner JB, Shorter RG (eds), 209-227, Philadelphia: Lea and Febiger.

Drossman D (1998) Presidential access: gastrointestinal illness and the biopsychosocial model, *Psychosomatic Medicine*, *60*, *258-267*.

Drossman D, Patrick DL, Mitchell CM, Zagami EA, Applebaum MI (1989) Health-related quality of life in inflammatory bowel disease: functional status and patients worries and concerns, *Digestive Diseases and Sciences*, 34, 1379-1386.

Drossman D, Leserman J, Mitchell CM, Li MZ, Zagami EA, Patrick DL (1991) Health status and health care use in persons with inflammatory bowel disease, *Digestive Diseases and Sciences*, 36, 1746-1755.

Fielding N (1993) Interviewing, in Gilbert N (ed.), Researching Social Life, London: Sage.

Ford AC, Moayyedi P, Hanauer SB (2013) Ulcerative colitis. BMJ, 346, f432

Fuller-Thomson E, Sulma J (2006) Depression and inflammatory bowel disease: findings from two nationally representative Canadian surveys. *Inflamm Bowel Dis*, 12, 697-707.

Gay M *et al.* (2011) *Crohn's, Colitis and Employment – from Career Aspirations to Reality,* Hertfordshire: Crohn's and Colitis UK. Available at <a href="http://www.chrohnsandcolitis.org.uk/Resources/CrohnsAndColitisUK/Documents/Crohns\_Colitis\_Employment.pdf">http://www.chrohnsandcolitis.org.uk/Resources/CrohnsAndColitisUK/Documents/Crohns\_Colitis\_Employment.pdf</a>. Accessed January 2014, p.13.

Goodhand JRN, Kamperidis NM, Joshi M, Wahed Y, Koodun EJ, Cantor NM, Croft *et al.* (2012) The phenotype and course of Inflammatory Bowel Disease in UK patients of Bangladeshi descent, *Alimentary Pharmacology & Therapeutics*, 35, 929-940.

Goodhand JR, Wahed M, Mawdsley JE, Farmer AD, Aziz Q, Rampton DS (2012) Mood disorders in Inflammatory Bowel Disease: relation to diagnosis, disease activity, perceived stress, and other factors, *Inflamm Bowel Dis*, 18, 12, 2301-2309.

Graff LA *et al.* (2009) Depression and anxiety in inflammatory bowel disease: a review of comorbidity and management, *Inflamm Bowel Dis*, 15, 7, 1105-1118.

Hall NJ, Rubin GP, Dougall A, Hungin APS, Neely J (2005) The fight for 'health-related normality': a qualitative study of the experiences of individuals living with established Inflammatory Bowel Disease, *Journal of Health Psychology*, 10, 3, 1 May, 443-455.

Hall NJ, Rubin GP, Hungin APS, Dougall A (2007) Medication beliefs among patients with Inflammatory Bowel Disease who report low quality of life: a qualitative study, *BMC Gastroenterology*, 7, 20.

Hanauer SB (2006) Inflammatory Bowel Disease: epidemiology, pathogenesis, and therapeutic opportunities, *Inflamm Bowel Dis*, 12, 1, S3-S9.

Häuser W *et al.* (2011) Anxiety and depression in patients with inflammatory bowel disease: comparisons with chronic liver disease patients and the general population, *Inflamm Bowel Dis*, 17, 2, 621-632.

Hjortswang H, Almer S, Strom M (1999) The network: a strategy to describe the relationship between quality of life and disease activity. The case of inflammatory bowel disease, *European Journal of Gastroenterology and Hepatology*, 11, 1099-1104.

Holloway W and Jefferson T (2000), *Doing Qualitative Research Differently: Free Association, Narrative and the Interview Method*, London: Sage.

The IBD Standards Group (2013) IBD Standards Update.

Jayaweera (2011) *Health of Migrants in the UK: What do we know?*, Migration Observatory Briefing, Oxford: COMPAS, University of Oxford.

Kalla R, Ventham NT, Satsangi J, Arnott IDR (2014) Crohn's Disease, *BMJ*, 349, Journal Article. doi:10.1136/bmj.g6670.

Karlsen S, Nazroo J (2006) Defining and measuring ethnicity and 'race': theoretical and conceptual issues for health and social care research, in Nazroo JY (ed.), *Health and Social Research in Multi-ethnic Societies*, London: Routledge.

Karlsen S, Becares L, Roth M (2012) Understanding the influence of ethnicity on health, in Craig G, Atkin K, Chattoo S, Flynn R (eds) *Understanding 'Race' and Ethnicity*, Bristol: The Policy Press.

Kelly M (1992) Colitis, London, New York: Routledge.

Kemp K, Griffiths J, Lovell K (2012) Understanding the health and social care needs of people living with IBD: a meta-synthesis of the evidence, [In eng], *World J Gastroenterol*, 18, 43, 21 Nov, 6240-6249.

Kurina LM, Goldacre MJ, Yeates D, Gill LE (2001) Depression and anxiety in people with inflammatory bowel disease, *Journal of Epidemiology and Community Health*, 55, 716-720.

Legal Status of Traditional Medicine and Complementary/Alternative Medicine: A Worldwide Review, World Health Organization (WHO). Accessed: Tuesday June 24, 2014), c.8.5.

Lerebours EC, Gower-Rousseau V, Merle F, Brazier S, Debeugny R, Marti JL, Salomez M. F, Hellot JL, Dupas JF, Colombel A, Cortot and Benichou (2007). Stressful life events as a risk factor for inflammatory bowel disease onset: A population-based case-control study, *Am J Gastroenterol* 102, 1, 122-131.

Lewis J, Ritchie J (eds) (2003) Qualitative Research Practices, London: Sage.

Loftus EV Jr. *et al.* (2011) Increased risks of developing anxiety and depression in young patients with Crohn's Disease, *Am J Gastroenterol*, 106, 9, 1670-1677.

Marri SR, Buchman AL (2005) The education and employment status of patients with Inflammatory Bowel Diseases, *Inflamm Bowel Dis*, 11, 2, 171-177.

Maunder R, Esplen MJ (1999) Facilitating adjustment to inflammatory bowel disease: a model of psychosocial intervention in non-psychiatric patients, *Psychotherapy and Psychosomatics*, 68, 230-240.

Mayberry JF, Farrukh, A (2012) Gastroenterology and the provision of care to Panjabi patients in the UK, *Frontline Gastroenterology*, <a href="http://fg.bmj.com/content/early/2012/04/12/flgastro-2012-100119.abstract">http://fg.bmj.com/content/early/2012/04/12/flgastro-2012-100119.abstract</a>

McCormick JB, Hammer RR, Farrell RM, Geller G, James KM, Loftus EV Jr., Mercer MB, Tilburt JC, Sharp RR (2012) Experiences of patients with chronic gastrointestinal conditions: in their own words, *Health Qual Life Outcomes*, 10, 25.

Mikocka-Walus AA, Andrews JM, Bernstein CN, Graff LA, Walker JR, Spinelli A, Danese S, van der Woude CJ, Goodhand J, Rampton D, Moser G (2012) Integrated models of care in managing inflammatory bowel disease: a discussion, *Inflamm Bowel Dis*, 18, 8, 1582-1587.

Miles MB, Huberman AM (1994) *Qualitative Data Analysis* (2<sup>nd</sup> edition), Thousand Oaks, CA: Sage.

Molodecky NA, Soon IS, Rabi DM, Ghali WA, Ferris M, Chernoff G, Benchimol EI *et al.*(2012) Increasing incidence and prevalence of the Inflammatory Bowel Diseases with time, based on systematic review, [In eng], *Gastroenterology*, 142, 1, Jan, 46-54. e42; quiz e30.

Moody GA, Bhakta P, Mayberry JF (1993) Disinterest in local self-help groups amongst patients with inflammatory bowel disease in Leicester, *International J of Colorectal Disease*, 8, 4, 181-183.

Moody GA, Eaden JA, Bhakta P, Sher K, Mayberry JF (1998) The role of complementary medicine in European and Asian patients with inflammatory bowel disease, *Public Health*, 112, 4, 269-271.

Moradkhani A, Beckman L, Tabibian JH (2013) Health-related quality of life in inflammatory bowel disease: psychosocial, clinical, socioeconomic, and demographic predictors, *Journal of Crohn's and Colitis*, 7, 6, 467-473.

Moser G (2006) Should we incorporate psychological care into the management of IBD? *Nat Clin Pract Gastroenterol Hepatol*, 3, 8, 416-417.

Mowat C, Cole A, Windsor A, Ahmad T, Arnott I, Driscoll R, Mitton S, Orchard T, Rutter M, Younge L, Lees C, Ho GT, Satsangi J, Bloom S (2011) IBD section of the British Society of Gastroenterology. Guidelines for the management of inflammatory bowel disease in adults, *Gut*, 60, 5, 571-607.

Mukherjee S, Sloper P, Turnbull A (2002) An insight into the experiences of parents with inflammatory bowel disease, *Journal of Advanced Nursing*, 37, 4, 355-363.

Nahon S, Lahmek P, Durance C, Olympie A, Lesgourgues B, Colombel J-F, Gendre J-P (2012) Risk factors of anxiety and depression in Inflammatory Bowel Disease, *Inflamm Bowel Dis*, 18, 11, 2086-2091.

Nash A, Lloyd M, Brooks F (2011) *Inflammatory Bowel Disease and Young People from Black and Minority Ethnic Communities in the UK,* Hertfordshire: CRIPACC, University of Hertfordshire.

Office for National Statistics (2012) *Religion in England and Wales 2011*, London: Office for National Statistics.

Office for National Statistics (2013) Comparison of mid-2010 Population Estimates by Ethnic Group against the 2011 Census, London: Office for National Statistics.

Norton C *et al.* (2013) Faecal incontinence in inflammatory bowel disease: associations and effect on quality of life, *Journal of Crohn's and Colitis*, 7, 8, e302-e311.

Patrick DL, Deyo RA (1989) Generic and disease specific measures in assessing health status and quality of life, *Medical Care*, 27, S217-S232.

Pope C, Ziebland S, Mays N (2000) Analysing qualitative data, *BMJ*, January 8, 320, 7227, 114–116.

Postnote (2007) Ethnicity and Health, Number 276, London.

Probert CS, Bhakta P, Bhamra B, Jayanth V, Mayberry JF (1996) Diet of South Asians with inflammatory bowel disease, *Arg Gastroenterology*, 33, 3, 132-135.

Ravikumara M, Sandhu BK (2006) Epidemiology of Inflammatory Bowel Diseases in Childhood, *Indian J of Paediatrics*, 73 August, 717-721.

Rochelle TL, Fidler H (2013) The importance of illness perceptions, quality of life and psychological status in patients with ulcerative colitis and Crohn's disease, *Journal of Health Psychology*, 18, 7, 972-983.

Royal College of Physicians (2014) *National Audit of Inflammatory Bowel Disease* (*IBD*) *Service Provision. UK IBD Audit*, London: Royal College of Physicians.

Sainsbury A, Heatley RV (2005) Psychosocial factors in the quality of life of patients with Inflammatory Bowel Disease, *Alimentary Pharmacology & Therapeutics*, 21, 499-508.

Saper RB, Phillips RS, *et al.* (2008) Lead, mercury, and arsenic in US- and Indian-manufactured medicines sold via the internet, *JAMA*, 300, 8, 915–923. doi:10.1001/jama.300.8.915. PMC 2755247. PMID 18728265.

Silverman D (2009) *Doing Qualitative Research: A practical handbook*, London: Sage.

Szczepura A (2005) Access to health care for ethnic minority populations, *Postgraduate Medical Journal*, 81, 141-147.

Tavernier N, Fumery M, Peyrin-Biroulet L, Colombel JF, Gower-Rousseau C (2013) Systematic review: fertility in non-surgically treated Inflammatory Bowel Disease, *Alimentary Pharmacology & Therapeutics*, 38, 8, 847-853.

Theis MK, Boyko EJ (1994) Patients perceptions of causes of inflammatory bowel disease, *American Journal of Gastroenterology*, 89, 1920.

Thompson RA (1990) Vulnerability in research: a developmental perspective on research risk, *Child Development*, 61, 1-16.

Warr D, Mann R, Tacticos T (2011) Using peer-interviewing methods to explore placed-based disadvantage: dissolving the distance between suits and civilians, *International J of Soc Research Methodology*, 14, 5, 337-352.

Williams JG, Roberts SE, Ali MF *et al.* (2007) Gastroenterology services in the UK. The burden of disease and organisation and delivery of services for the gastrointestinal and liver disorders: a review of the evidence, *Gut*, 56, 1-113.

Wilson (2011) Establishment of the UK & Irish IBD Genetics Consortium, http://www.nacc.org.uk/downloads/research/2011/DavidWilson.pdf.

Wolfe BJ, Sirois FM (2008) Beyond standard quality of life measures: the subjective experiences of living with inflammatory bowel disease, *Qual Life Res*, 17, 6, 877-886.

# **Appendix 1** Interview Topic Guide

# 1. Introduction (10 mins MAX)

**Briefing note:** This section covers the preliminaries to the interview itself. Use this time to get a 'feel' for the interviewee (and them for you!), establish rapport, ascertain any language issues. The section concludes with working through the consent form. Whilst necessary, it is important not to make completing the form too 'formal' or intimidating. By the end of this section you should have either switched on the recorder or be ready to

- Thank for interest in project and time given to take part.
- Introduce self
  - interviewer working for University of York, carrying out a research project funded by Crohn's and Colitis UK.
  - o Explain that not medically trained
  - Project is working with 5 gastroenterology clinics in different parts of the country. By the end of the project will have interviewed ~35 IBD patients
- Remind about the project and purpose of interview:
  - to find out about people's experiences of living with IBD and the types of support they would like from Crohn's and Colitis UK and from health services.
     These are the topics the interview will cover.
  - Purpose is to pass on recommendations to Crohn's and Colitis UK & Gastroenterology clinics as to how they can improve the services they provide
  - REMIND/EMPHASISE: they can ask to skip any questions they don't want to answer and/or end the interview anytime they like.
- Check re timings
  - Explain that interview will take around an hour (maybe a little longer).
  - Check how much time they have available, any interruptions anticipated.
  - o Explain that they can take a break whenever they wish.
- Recording the interview.
  - Explain: it's easier for us to audio-record but, if not happy can make notes instead. It's their choice.
  - Reassure:
    - all data and information collected is treated as confidential.
    - it is only seen by members of the research team.
    - it is very stored securely.

- Check if have any questions
- Check if happy to proceed
- If given permission, turn recorder on
- Complete consent form
- If you don't have any background information on the person, explain that at present all you know about them is what they have written on the response, form, so before you begin the interview it would be helpful to know a little more about their circumstances. Then check:
  - Whether working, studying or caring for others
  - Who they currently live with

# 2. Diagnosis, management and other's reactions (~ 15 mins)

**Briefing note:** There is <u>no need</u> to obtain a detailed history/description of symptoms and treatments. Early on you will identify what words the interviewee uses to refer to/describe their condition. Use these words during the remainder of the interview. Here we are also trying to find out about their understanding of the condition/ treatments. We also explore

- Explain that at present all you know about their health is what they have written on the response, form, so would like to begin by finding out a little bit about their health.
- Brief history
  - Check specific diagnosis, and when diagnosed
- Current health/situation with condition
  - How has your health been over the past year? (IBD; minor illnesses; tiredness/energy levels)
  - Frequency of flare-ups (eg. approx no times/year; % of year badly affected.

## Management

- Past and present treatments:
  - medication, diet,...anything else
     (NB Make a note of any medications as you will need this information at the end of the interview to complete 'Background Information Form 2'.)
  - understanding of how current treatments manage the condition
  - any issues / difficulties /concerns sticking to treatment regime
- Any surgery

- very brief details; successful or not
- o Have you declined any medical treatments or investigations?
  - explore reasons
- Any alternative/non NHS treatments/approaches to managing condition
  - brief details and reasons for pursuing
  - clinic awareness of & reaction to use of these treatments
- o Any ways in which your faith influences how you manage your condition
  - We know some people use faith healers. Have you?
    - Brief details and reasons for pursuing

# Understanding of condition

- o Reflect back on how much they appear to know about IBD, THEN
  - Check where they gathered this knowledge from
  - check if there is anything unsure about and would like more information on

# Family's reactions to the investigations and treatments

- If you don't already know, check family composition & who is in the household.
- o Who in close and wider family knows about diagnosis
  - justification for who have told/not told
- How have the responded to diagnosis and, treatments and lifestyle recommendations
  - check for both positive & negative responses
- Who have you decided not to tell and why

# Other's knowledge/ reactions

- Who knows else have you talked to about having IBD
  - justification for who have told/not told
  - their reactions

# 3. Living with IBD (~15 minutes)

**Briefing note:** In this section we want to hear whether/how IBD affects people's lives. Because IBD is a fluctuating condition, it is important that you emphasise that we would like them to think about when things are bad (a flare up) and when things are more settled.

We have created a visual aid ('life map') to facilitate this section in case people find it difficult to spontaneously identify/talk about the impact of IBD on their lives.

This section ends with exploring who supports/looks after them when they are poorly with IBD in terms of family/informal support networks. The next part of the interview will explore support from the clinic.

- o For each distinct issue mentioned, probe using questions such as:
  - Can you explain a little more how it affects ......
  - Can you give me an example of when that has happened?
- Show the interviewee the 'life map'. Some areas may have already been mentioned.
  - Ask interviewee to identify any further life areas affected by having IBD. Use same pattern of questioning as above.
  - o Repeat until feel have explored as much as interviewee wants to discuss.

## Support when unwell

- O Who tends to know?
- O Who supports / helps out when unwell?
- Ask if feel adequately supported.
- Ask to describe any areas where need more help/support

## Reflections on being South Asian

- This project is about the experiences of people are from South Asian communities. We'd really like to know whether living with IBD is different in any way for people from this background. What are your views on this? Does being [use ethnic group option ticked in response form] and/or [use faith ticked in response form] make any difference to the impact IBD has had on your life?
- If <u>not</u> made a difference to them, check if they feel it makes a difference to other people who are South Asian makes a difference to the Impact IBD has on *other* people's lives

# 4. Contact with gastroenterology services (~15 mins)

Briefing note: In this section we want to find out about their experiences of NHS clinics / other services. Patients may contact the clinic outside of routine appointments if they are concerned about their health, though not all do this. We also want to explore whether anyone accompanies them to appointments, the reason/role of that person (e.g. companion, chaperone, interpreter etc.) and feelings about the person's presence. Finally, we would like to find out about any inpatient experiences, specifically any difficult/negative or positive experiences of being on the ward. Across all these topics, we want to find out what would improve their experiences as NHS patients.

- Explain that this section is about experiences with clinics. Remind participant that your independent from the clinic everything they say will be kept confidential.
- Frequency of appointments at the clinic and who usually sees (doctor, nurse, dietician....)
- Ask if usually go to appointments on own or take someone with them
  - If take someone, explore why.
- Contact with clinic outside of appointments
  - Frequency and reasons for contact
  - How make contact (eg phone)
  - o Who speaks with?
- Check if had any overnight stays in hospital for IBD.
- Satisfaction and concerns ......
  - Anything particularly stand out about experiences with gastroenterology services: either a very positive or very difficult experience.

Check for

- Clinic appointments
- Contact between clinic appointments
- Inpatient care
- Perception of whether clinic staff are aware of and respond to needs as a South Asian woman/man [insert words as appropriate]

Make sure to check for **both** positive and negative experiences

# 5. Support from Crohn's and Colitis UK (~15 mins)

**Briefing note**: This section of the interview is concerned with interviewee's knowledge and use of Crohn's and Colitis UK; and views on accessing support from such an organisation. We use a 'map' of the services/support Crohn's and

- Signal that this is the last section of the interview (as interviewees may begin to tire at this stage)
- Explain that Crohn's and Colitis UK have funded the research because they are concerned that what they provide is not appealing to people from South Asian communities, so it would be very helpful to have their views on what they provide.
- Awareness of/use of Crohn's and Colitis UK and the support / services they offer.

- Show list of support and services offered by Crohn's and Colitiis UK
- Check for any contact with / use of Crohn's and Colitis UK resource
  - Reasons for contact / use of resource
  - Views on quality of support / information.
- o Is there anything on the list that you didn't know they provided, which would have been useful in the past if you had known about it or which you might be interested in using in the future?
- Identify which support/services they regard as <u>not</u> useful/ of interest to them
  - Why not
- What could Crohn's and Colitis UK do to:
  - o increase the number of people who are members of their organisation?
  - o cater for the needs of South Asian patients?

# 6. Close

**Briefing note**: This is the final section of the interview. It provides interviewees with a final chance to raise issues/mention views. Thanks for participation is <u>very important</u>, as is reminding about confidentiality.

We close with some practical aspects of participation.

- Any further ideas and recommendations for improving support to people of South Asian origin and their families
  - NHS.
  - Crohn's and Colitis UK.
  - Other
- Thank participants for their time and thoughts
- Explain that before end the interview need to check some background information on participants. **Complete 'Background Information Form 2'.**
- Explain that will be interviewing people until end of October, and will then analyse all
  the data and write a report. This work needs to be completed by April 2014 as this is
  when the project ends. A summary of the findings will then be available through the
  SPRU website. Ask if they would also like to be sent a copy of the findings through
  the post. (Record this on the background info form)
- Check if they have any further questions.

- 'Useful Contacts' leaflets
  - o For face to face interviews hand out leaflet
  - o For telephone interviews mention that will be putting this leaflet in the post to them along with voucher.
- Find out about voucher preference and explain that it will be sent in the post, along with a receipt which would like them to sign so that we know they have received it.

# **Appendix 2** Visual Aids for Interview

# Membership

- Newsletter
- 'Can't wait' card
- Vote at AGM

# **Local support groups**

- Meetings
  - Informal meetings
  - On IBD related topics
- Local newsletter

# **Telephone helplines**

- Information & advice
- Emotional support
- Disability benefits service

# Crohn's and Colitis UK



# **Financial grants**

- Bedding
- Clothes
- Washing
- Educational

# Teenagers & young adults

- Information leaflets
- On line forum
- Summer camp
- Workshops

# Leaflets and booklets

- Understanding IBD
- Managing symptoms
- Travelling abroad
- Explaining IBD to children

- Treatments
- Living with IBD
- Disability benefits
- Employment

# Leaflets available in Asian languages:

- 'Living with IBD'
- 'Understanding Colitis & Crohn's Disease'

My relationship with my partner / wife / husband

Family life

Getting out & about, travelling

My feelings & emotions

Being a parent

Living with IBD

Family or community gatherings/events

Work or studying

Practising my faith

Social activities with friends

My hopes & plans for the future

Cooking, housework and/or jobs around the house

# **Appendix 3** The development of the thematic framework

Development of the thematic framework involved a number of steps as follows:

- 1. The lead researcher (R1) carried out an initial read through of all transcripts and field notes and then wrote a 'pen portrait' for each interview. A pen portrait is a 1-2 page summary which 'which makes the person come alive to the reader' (Holloway and Jefferson, 2000). It includes any inconsistences, contradictions and puzzles noticed within the data. In addition, the researcher developed a list of all the themes observed in each interview.
- 2. Next the R1 passed on a sample of five transcripts, representing participants of different genders, faiths, ages, and years of experience of living with IBD to two members of the research team (R2 and R3). Having read these transcripts, the research team met to discuss: (1) the *A priori* and emergent themes identified within the data and (2) whether any of these themes appeared to be grounded in South Asian identity or culture. At these stage it was agreed that the data fell into 8 higher order themes (The impact of IBD on people's lives; strategies that help people live with IBD; understanding of IBD; disclosing illness to others and their response; going abroad; experiences with health services; unmet support needs; feedback and suggestions for Crohn's and colitis UK).
- 3. R1 reviewed all the themes identified from the initial read through of the data and the team meeting, and assigned them to one of 8 higher order themes. This constituted the first draft of the thematic framework.
- 4. This first draft of the thematic framework was then presented to the PAG for comment. Feedback was as follows:
  - a. The PAG validated a number of themes emerging from the data (pain, fatigue, being housebound), but noted that many of these were generic and asked that the final report makes clear which experiences are likely to be specific to South Asian patients.
  - b. They note that a number of beliefs about the cause of IBD seemed to be about self-blame.
  - c. Diet was not as prominent an issue as expected. Members of the PAG were interested in understanding which dietary restrictions were particularly difficult for South Asian patient and why, and asked that this is addressed in the final report.

- d. The PAG was interested in the importance of people's living circumstances in shaping their experiences and suggested that it might be worth exploring differences between those who do and do not live with extended family.
- e. Patient representatives *not* of South Asian background urged caution in labelling experiences as South Asian, reporting that some of the themes which might appear to be linked to South Asian faith or culture were also familiar to them (e.g. IBD being a punishment for sins).
- 5. R1 and R2 met to revise thematic framework in light of feedback from PAG. The following decisions were made:
  - a. R1 & R2 reflect on the fact that while the PAG were familiar with the idea that pain and fatigue are central to IBD patients experiences and quality of life, they seemed less familiar with the possibility that anxiety about loss of bowel control can be significant for some people's lives, causing them to be housebound even when their symptoms are well controlled. This is seen as important by the research team as it suggests that improvements to quality of life could be brought about by some form of psychological intervention.
    Efforts will be made to explore whether anxiety is as prominent within the data as it appeared to be during the first read through.
  - b. In order to remove duplication of data being charted into different themes, it was decided that the themes 'Support needs expressed by patients' and 'Strategies people use to live with IBD' should be subsumed within 'The impact of IBD on people's lives'. It was also agreed that this newly merged theme should be given a different title as it is now about more than just 'impact' and forms the core of the interview. It will therefore be renamed 'Living with IBD'. Following this decision, the framework includes 6 themes: Living with IBD; Disclosure of IBD diagnosis to others & their response; Perceptions and beliefs about cause of IBD; Visits to South Asian countries; Experiences with health services; Feedback and recommendation for Crohn's and Colitis UK.
- 6. Next R1 carried out data extraction on 10 interviews, then met with R2 to review progress. After jointly reviewing the charts it is agreed that:
  - a. The themes 'Disclosure of IBD diagnosis to others and their response' and 'Experiences with health services' are both about support, so should be managed under one 'Social support' theme.

- b. Care needs to be taken with data on impact to be clear whether participant is referring to impact when having a flare up or on a more ongoing basis.
- 7. The final draft of the thematic framework (see Appendix X) is agreed. It includes 5 main themes: Living with IBD; Access to support; Perceptions and beliefs about the causes of IBD; Visits to South Asian countries; Feedback and recommendations to Crohn's and Colitis UK.

# **Appendix 4** Thematic framework

# Theme 1 - Living with IBD

#### Codes

- Getting diagnosed
- Symptoms
  - o Pain
  - Fatigue
  - Bowel control
- Anticipatory anxiety re bowel control
- Strategies for dealing with symptoms & any barriers to using them
  - Medication
  - o Surgery
  - Altering diet
  - Alternative/complementary therapies
  - Spiritual healer
  - o Lifestyle changes (i.e. rest, exercise, stress reduction etc.)
  - o Turning to faith
  - o Other
- Impact on life (positive or negative)
  - o Being confined to bed
  - Work/study
  - Financial
  - Fulfilling cultural expectations of good wife/husband/mother/father (including worries for future ability to take on role)
  - Activities of daily living (i.e. cooking, cleaning, self-care)
  - Social events/activities
  - o Caring responsibilities
  - o Relationships with spouses/partners
  - o Relationships with children
  - o Relationships with other family members
  - Friendships
  - o Religious practice
- Emotional well-being
- Outlook on life
- Articulated support needs

#### Theme 2 – Access to support

#### Notes:

This theme will include unsupportive responses, such as:

- lack of support in adhering to preferred diet/treatment
- unwanted advice i.e. suggestions to seek out alternative treatment

•

Barriers related to lack of knowledge will include any explanations as to why they have chosen/not to disclose.

#### Codes

- · Support in household
  - Partner
    - Perception of support from partner
    - Barriers to support (including lack of knowledge/understanding)
    - Facilitators/enablers to support

#### Parents

- Perception of support from parents
- Barriers to support (including lack of knowledge/understanding)
- Facilitators/enablers to support

#### Children

- Perception of support from children
- Barriers to support (including lack of knowledge/understanding)
- Facilitators/enablers to support

#### Other family members

- Perception of support from other family members in household
- Barriers to support (including lack of knowledge/understanding)
- Facilitators/enablers to support
- Contact with GP
- Articulated support needs

#### • Support from outside household

- Employers/work colleagues/academic supervisors
  - Perception of support
  - Barriers to support (including lack of knowledge/understanding)
  - Facilitators/enablers to support

#### Extended family and those in wider community

- Perception of support
- Barriers to support (including lack of knowledge/understanding)
- Facilitators/enablers to support

#### Friends

- Perception of support
- Barriers to support (including lack of knowledge/understanding)
- Facilitators/enablers to support
- The web
- Articulated support needs

# Support from health

- Gastroenterology
  - Perception of support from Gastroenterology
  - Barriers to support (including lack of knowledge/understanding)
  - Facilitators/enablers to support

#### Inpatient services

- Perception of support from inpatient services
- Barriers to support (including lack of knowledge/understanding)

Self-blame

- Facilitators/enablers to support
- o Contact with GP
- Articulated support needs

#### Theme 3 – Perceptions and beliefs about cause of IBD

## Codes

- Cause unclear
- Cause Anxiety/stress
- Cause Lifestyle
- Cause diet
- Cause punishment
- Cause God/Allah's will
- Cancer related
- Articulated support needs

# **Theme 4 - Visits to South Asian countries**

# Codes

- Purpose of visit
- Impact of visit on health
- Adjustments made to medical treatment when abroad
- Seeking further/alternative treatment when abroad
- Barriers to getting support/treatment when abroad
- Articulated support needs

# Theme 5 - Feedback & recommendations to CCUK

#### Codes

- Awareness of CCUK
- Helpful support
- Unhelpful support
- Ways in which support could be developed generic
- Ways in which support could be developed for South Asian patients